

**BOARD OF COUNTY COMMISSIONERS** 

Public Services Building2051 Kaen Road | Oregon City, OR 97045



## <u>Thursday, July 7, 2016 - 10:00 AM</u> BOARD OF COUNTY COMMISSIONERS

Beginning Board Order No. 2016-70

CALL TO ORDER

- Roll Call
- Pledge of Allegiance

**I.** <u>CITIZEN COMMUNICATION</u> (The Chair of the Board will call for statements from citizens regarding issues relating to County government. It is the intention that this portion of the agenda shall be limited to items of County business which are properly the object of Board consideration and may not be of a personal nature. Persons wishing to speak shall be allowed to do so after registering on the blue card provided on the table outside of the hearing room prior to the beginning of the meeting. Testimony is limited to three (3) minutes. Comments shall be respectful and courteous to all.)

**II.** <u>PUBLIC HEARING</u> (The following items will be individually presented by County staff or other appropriate individuals. Persons appearing shall clearly identify themselves and the department or organization they represent. In addition, a synopsis of each item, together with a brief statement of the action being requested shall be made by those appearing on behalf of an agenda item.)

 Board Order to Proceed with Construction of a Local Improvement District (LID) for Street and Storm Drainage Improvements on Starview Lane (Ken Kent, Department of Transportation & Development, Engineering)

**III.** <u>CONSENT AGENDA</u> (The following Items are considered to be routine, and therefore will not be allotted individual discussion time on the agenda. Many of these items have been discussed by the Board in Work Sessions. The items on the Consent Agenda will be approved in one motion unless a Board member requests, before the vote on the motion, to have an item considered at its regular place on the agenda.)

#### A. Health, Housing & Human Services

- 1. Approval of a Renewal Revenue Intergovernmental Agreement with the City of Lake Oswego for Medical Direction for the Fire Department and Communications Center – Public Health
- 2. Approval of Amendment #3 to the Intergovernmental Agreement with Multnomah County, for a Public Health Officer *Public Health*
- 3. Approval of a Renewal Grant Agreement with the US Department of Housing and Urban Development, Continuum of Care Program, for the Homeless Management Information System *Housing & Community Development*

## B. Elected Officials

1. Approval of Previous Business Meeting Minutes – *BCC* 

#### IV. WATER ENVIRONMENT SERVICES

(Service District No. 1, Tri-City Service District & Surface Water Management Agency of Clackamas County)

1. Board Order Authorizing Refund Revenue Bonds to Reduce Debt Service Expense for Clackamas County Service District No. 1

## V. COUNTY ADMINISTRATOR UPDATE

## VI. COMMISSIONERS COMMUNICATION

NOTE: Regularly scheduled Business Meetings are televised and broadcast on the Clackamas County Government Channel. These programs are also accessible through the County's Internet site. DVD copies of regularly scheduled BCC Thursday Business Meetings are available for checkout at the Clackamas County Library in Oak Grove. You may also order copies from any library in Clackamas County or the Clackamas County Government Channel. www.clackamas.us/bcc/business.html



M. Barbara Cartmill Director

#### DEPARTMENT OF TRANSPORTATION AND DEVELOPMENT

DEVELOPMENT SERVICES BUILDING 150 BEAVERCREEK ROAD OREGON CITY, OR 97045

July 7, 2016

Board of County Commissioners Clackamas County

Members of the Board:

## Board Order to Proceed with Construction of a Local Improvement District (LID) for Street and Storm Drainage Improvements on Starview Lane

Purpose/Outcomes	Approval to proceed with bidding and construction of the Starview Lane Local Improvement District.
Dollar Amount and Fiscal Impact	The preliminary cost estimate for the proposed LID is \$49,251. If approved, construction is anticipated late summer of 2016. This project is currently not budgeted.
Funding Source	LID Construction Fund, paid for by a special assessment against the benefitting properties, with reimbursement from property owners through lump sum payment or financing over 10 years.
Duration	Remains in effect for two years. If improvements are not completed within two years, the Board may adopt an order vacating the LID.
Previous Board Contact	5/26/16 BCC Business Meeting – Order No. 2016-43 Accepting the Feasibility Report for the Starview Lane LID and directing DTD Engineering to mail to each property owner of each parcel of land to be assessed for the improvements, a written notice of the favorable report, the estimated cost of the improvement and the estimated amount of assessment against each parcel.
Strategic Plan Alignment	Build a strong infrastructure.
Contact Person	Kenneth Kent, DTD Engineering, Senior Planner 503-742-4673
Contract No.	None

## **BACKGROUND:**

The Starview Lane LID was initiated by the Board with adoption of Board Resolution 2016-22 on February 25, 2016, based on a petition signed by 94.73% of the owners, representing 92.50% of the land abutting the proposed Starview Lane improvements. The Preliminary Feasibility Report for this project was accepted by the Board on May 26, 2016 with adoption of Board Order 2016-43. As a result of that action, all owners were notified of their estimated assessment, and were provided a 20-day remonstrance period to submit written objection to formation of the LID. No remonstrances were received during the 20-day period.

According to ORS 371.630, remonstrances are required from more than fifty percent (50%) of the land owners as minimally necessary to declare the project abandoned. Because of the high level of support, and the number of remonstrances did not exceed fifty percent (50%), staff will recommend that this LID be approved to proceed.

The preliminary cost estimate for the proposed LID is \$49,251. Capital improvement costs of this LID will be the responsibility of the benefited property owners. Costs of these improvements will be paid for by a special assessment against the properties benefited. Financing for the project will be obtained through the Finance Director using the LID Construction Fund, with reimbursement from property owners through lump sum payment or financing over 10 years.

Proposed LID assessments are based on a division of the road improvement costs into three sections, A, B and C, based on length of road travelled. The estimated assessments per property are \$1,549.59, \$2,212.29 and \$2,704.80, with properties at the beginning of the road paying the lower amounts.

Starview Lane is a local access road that is not maintained by the county. With completion of the proposed road improvements, Starview Lane will remain a local access road that is not maintained by the county.

With the Board's approval to proceed with the LID, subsequent steps in the process will include:

- Staff proceeds with bidding and construction, anticipated to occur toward the end of this summer.
- Following completion of construction, a final report will be submitted to the Board identifying the final project cost and assessment amount per property. The Board will take public testimony regarding the final assessment, and if approved, the Board will adopt a resolution imposing the final assessments.

This Order has been reviewed and approved by County Counsel.

#### **RECOMMENDATION:**

Staff recommends that the Board approve this order which, in accordance with ORS 371.635, will direct this Department to proceed with the improvement by contract.

Respectfully submitted,

Kenneth Kent, DTD Engineering, Senior Planner

In the Matter of Proceeding to Construct a Local Improvement District by Order of Order No. the Board of County Commissioners for Street and Storm Drainage Improvements Page 1 of 2 to Starview Lane in Sections 12B and 12C, T3, R3E, W.M., Clackamas County, Oregon

This matter coming before the Clackamas County Board of Commissioners at the regularly scheduled public meeting on July 7, 2016 to consider approval to proceed with construction of the proposed street and storm drainage improvements for Starview Lane, located in Sections 12B and 12C, T3, R3E, W.M., Clackamas County, Oregon; and

**WHEREAS**, a petition has been received requesting a Local Improvement District be formed for Starview Lane; and, said Petition contains signatures of 94.73% of the owners, representing 92.50% of the land abutting the improvements; and

**WHEREAS**, by Resolution No. 2016-22, in accordance with Clackamas County Code Chapter 4.02, and ORS 371.615, the Board did initiate the Local Improvement District; and

**WHEREAS**, by Resolution No. 2016-22, the Department of Transportation and Development, in accordance with Clackamas County Code Chapter 4.02, and ORS 371.625, was directed to prepare a feasibility and cost report and recommend a method of assessment; and

**WHEREAS**, in accordance with ORS 371.625, that the Director of the Department of Transportation and Development finds the proposed improvement feasible, has reported the estimated cost of said improvement and recommended a method of assessment to the Board; and

**WHEREAS**, by Order No. 2016-43, in accordance with ORS 371.625, did adopt a method of assessment for the improvement of Starview Lane; and

**WHEREAS**, by Order No. 2016-43, in accordance with ORS 371.630, did direct that written notices be mailed to all affected property owners, notifying them of the favorable report, the estimated cost of the improvement and the estimated amount of the assessment against the land of the owner, and calling for remonstrances against said project; no remonstrances were received.

NOW, THEREFORE, the Clackamas County Board of Commissioners do hereby order: that pursuant to the provisions of ORS 371.635, Starview Lane, a public road

In the Matter of Proceeding to Construct a Local Improvement District by Order of Order No. the Board of County Commissioners for Street and Storm Drainage Improvements Page 2 of 2 to Starview Lane in Sections 12B and 12C, T3, R3E, W.M., Clackamas County, Oregon

located in Sections 12B and 12C, T3, R3E, W.M., Clackamas County, Oregon, be improved by contract; said improvements to include as follows:

Road repair of sections of the asphalt surface and aggregate base Shoulder repair, ditch cleaning, and drainage improvements; and

**IT IS FURTHER ORDERED**, that all properties as shown in the Preliminary Feasibility Report, and on Exhibit "A", attached hereto, in accordance with ORS 371.635, be subject to a lien on each subject property's assessment for the cost of the improvement, in an amount to be determined later by order of the Board of County Commissioners; and.

**IT IS FURTHER ORDERED**, that this Order, in accordance with ORS 371.635, shall be recorded and indexed in the Assessment Lien Docket in the Office of the County Clerk.

DATED this \_\_\_\_\_ day of \_\_\_\_\_, 2016.

CLACKAMAS COUNTY BOARD OF COMMISSIONERS

Chair

Recording Secretary

## Exhibit "A" STARVIEW LANE LOCAL IMPROVEMENT DISTRICT ESTIMATED ASSESSMENT

## Assessment Rates

TOTAL ESTIMATED COST	\$49,251.00
NUMBER OF SHARES	19
COST/SHARE Section A	\$1,549.59
COST/SHARE Section A+B	\$2,212.29
COST/SHARE Section A+B+C	\$2,704.80

PROPERTY OWNER	TAX LOT	MARKET VALUE	ASSESSABLE SHARE	TOTAL EST ASSESSMENT
ANDERSON ARNOLD W & SANDRA J	32E12C 01700	\$310,770	A	\$1,549.59
19623 S CREEK ROAD	J2L12C 01700	\$510,770	~	φ1,349.39
OREGON CITY, OREGON 97045				
WILSON BRETT	32E12B 02702	\$145,269	A+B	\$2,212.29
19496 S STARVIEW LANE	522120 02,02	¢,		ΨΖ,ΖΤΖ.ΖΟ
OREGON CITY, OREGON 97045				
MELLIS JAMES	32E12C 01801	\$266,201	A+B	\$2,212.29
19505 S STARVIEW LANE				
OREGON CITY, OREGON 97045				
SORENSEN PAUL J & SUSAN M	32E12B 03501	\$445,252	A+B+C	\$2,704.80
19300 S STARVIEW LANE				
OREGON CITY, OREGON 97045				
GARFIELD SCOTT D CO-TRUSTEE	32E12B 03400	\$363,383	A+B+C	\$2,704.80
19322 S STARVIEW LANE				
OREGON CITY, OREGON 97045		•		<b>1</b>
WILLIAMSON DALE A	32E12B 03301	\$371,794	A+B+C	\$2,704.80
19333 S STARVIEW LANE				
OREGON CITY, OREGON 97045	225425 02200	<b>#040.044</b>		<b>#0.704.00</b>
SPEASL JUSTIN A & TABITHA B	32E12B 03300	\$343,644	A+B+C	\$2,704.80
19344 S STARVIEW LANE OREGON CITY,OREGON 97045				
CHESS JILL	32E12B 02902	\$201,713	A+B+C	¢0.704.90
19388 S STARVIEW LANE	32E12D 02902	φ201,713	A+D+C	\$2,704.80
OREGON CITY, OREGON 97045				
ANDERSON LEONARD J & DEBBIE	32E12B 02700	\$400,082	A+B+C	\$2,704.80
19440 S STARVIEW LANE	522120 02/00	\$100,00 <u>2</u>	AIDIO	ψ2,704.00
OREGON CITY, OREGON 97045				
BOYCE NICHOLAS G & KIMBERLY M	32E12B 03003	\$119,001	A+B+C	\$2,704.80
19464 S STARVIEW LANE				, ,. <b>. </b>
OREGON CITY, OREGON 97045				
LAWRENCE RUSSELL A & CAROL A	32E12B 02703	\$460,793	A+B+C	\$2,704.80

19478 S STARVIEW LANE OREGON CITY,OREGON 97045				
TRACY PENNY M 19416 S STARVIEW LANE OREGON CITY,OREGON 97045	32E12B 02902	\$328,489	A+B+C	\$2,704.80
SMITH CASEY LEE & JULIE LEE 17001 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12B 03502	\$188,864	A+B+C	\$2,704.80
GALLAGHER JOHN & KIM H 17041 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01902	\$389,497	A+B+C	\$2,704.80
PAULSON LON & CHARLENE 17100 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01900	\$523,537	A+B+C	\$2,704.80
HAMBERG KURTIS & JOY 17105 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01903	\$522,591	A+B+C	\$2,704.80
SIMMONS KEITH 17171 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01904	\$326,408	A+B+C	\$2,704.80
KUYKENDALL DALE A & DIANNE M 17200 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01905	\$493,135	A+B+C	\$2,704.80
HODGE DALE 17223 S WINTERVIEW LANE OREGON CITY,OREGON 97045	32E12C 01901	\$543,648	A+B+C	\$2,704.80

Totals: (Estimated Project Cost \$49,251)

\$6,744,071 19 \$49,251



July 7, 2016

Board of County Commissioner Clackamas County

Members of the Board:

Approval of a renewal Revenue Intergovernmental Agreement with the City of Lake Oswego, for Medical Direction for the Fire Department and Communications Center

Purpose/Outcomes	This Agreement provides the basis for a cooperative working relationship
	for the provision of medical direction for the Lake Oswego Fire Department
	(LOFD) and Lake Oswego Communications Center (LOCOM).
Dollar Amount and	Contract maximum value is \$12,000.
Fiscal Impact	
Funding Source	Emergency Medical Services Coordination – No County General Funds will
	be used.
Duration	Effective July 01, 2016 and terminates on June 30, 2017
Strategic Plan	1. Improved community safety and heath
Alignment	2. Build a strong infrastructure
Previous Board	The Board of County Commissioners previously reviewed this agreement
Action	on June 26, 2013 agenda item 062913-A10 ,June 05, 2015 agenda item
	060514-A2, and June 25, 2015 agenda item 062515-A4
Contact Person	Dawn Emerick, Public Health Director – 503-655-8479
Contract No.	7816

#### BACKGROUND:

The Clackamas County Public Health Division (CCPHD) of the Health, Housing & Human Services Department requests the approval of an Agreement with the City of Lake Oswego to provide Medical Direction for the Fire Department and Communications Center. This Agreement provides the basis for a cooperative working relationship for the provision of medical direction for the Lake Oswego Fire Department (LOFD) and Lake Oswego Communications Center (LOCOM) such as, developing a program to ensure LOFD meets the state requirements and to establish performance standards. This agreement will ensure that LOFD first responders meet requirements and protocols for the provision of EMS care.

The maximum contract value is \$12,000. This contract is effective July 1, 2016 and continues through June 30, 2017.

#### **RECOMMENDATION:**

Staff recommends the Board approval of this contract and authorizes Richard Swift, H3S Director to sign on behalf of Clackamas County.

Respectfully submitted,

Richard Swift, Director Health, Housing, and Human Services

#### INTERGOVERNMENTAL AGREEMENT BETWEEN CLACKAMAS COUNTY AND THE CITY OF LAKE OSWEGO Contract # 7816

## 1. Purpose

- A. This Agreement is entered into between the Clackamas County (County) and the City of Lake Oswego (City) for the cooperation of units of local government under the authority of ORS 190.010.
- B. This Agreement provides the basis for a cooperative working relationship for the provision of medical direction for the Lake Oswego Fire Department (LOFD) and Lake Oswego Communications Center (LOCOM) and Lake Oswego Police Department (LOPD).

## 2. Scope of Cooperation

- A. County agrees to:
  - 1. Assign a mutually agreed upon physician to provide Medical Director Services to LOFD and LOCOM.
  - 2. Meet with LOFD personnel on a mutually agreed upon schedule to develop a program to:
    - Ensure that LOFD EMS providers meet Oregon State requirements for licensure and have the knowledge, skills and abilities to perform at the standards determined jointly by County and LOFD.
    - b. Evaluate each EMS Provider's skill performance annually.
    - c. Provide case reviews.
    - d. Oversee and direct training courses.
    - e. Oversee and direct a quality improvement program.
  - 3. Provide medical direction and approval of Priority Dispatch Cards and case reviews for LOCOM dispatchers.
  - 4. Oversee the maintenance, use, and documentation of all Automatic External Defibrillators (AEDs) provided for use by the City of Lake Oswego, in accordance with Federal and State regulations.
  - 5. Provide contact information so that LOFD personnel can contact assigned Medical Director (or designee) in a timely manner.
- B. City agrees to:
  - 1. Meet with County personnel on a mutually agreed upon schedule to

develop and maintain a program to:

- a. Ensure that LOFD EMS providers meet Oregon State requirements for licensure and have the knowledge, skills and abilities to perform at the standards determined jointly by County and LOFD.
- b. Evaluate each EMS Provider's skill performance annually.
- c. Provide case reviews.
- d. Oversee and direct training courses.
- e. Oversee and direct a quality improvement program.
- 2. Provide an EMS Coordinator to:
  - a. Coordinate training exercises and skill monitoring.
  - b. Maintain a computerized CQI database of all procedures and relevant training for all EMS providers.
  - c. Coordinate case reviews and necessary training for LOCOM dispatchers.
  - d. Provide periodic reports to guide training efforts.
  - e Organize the classes and locations, obtain instructors, and will notify Medical Director at least two (2) months in advance of the class as to Medical Director's role in said courses.
- 3. City further agrees to the following regarding the authority of the Medical Director:
  - a. The City will not permit its EMS Providers to practice at a level other than that approved by Medical Director.
  - b. LOFD personnel will not practice under the medical direction or protocol of any physician other than the one assigned by mutual agreement with the exception of on-line medical control or direct in-person physician supervision provided during patient encounters.
  - c. As per ORS 682-245, Medical Director has the final decision with respect to the standing orders and written authorization to provide EMS care by LOFD Department personnel.
  - d. Medical Director may require specific remedial action to correct deficiencies noted in the continuous quality improvement process, or identified violations of federal, state and local laws or regulations.
  - e. County is not an employer of its EMTs, and acknowledges that no employment relationship exists between County and the EMTs employed by the City.

## 3. Compensation

A. City will pay to County an amount not to exceed \$ 12,000.00 for services described in section 2A. Payments shall be requested and made as follows:

Payment of \$1,000.00 will be requested monthly by invoice from County. Payment will be made by City within 30 days of receipt of invoice.

B. All checks shall be made payable to Clackamas County and mailed to the following address:
 Clackamas County Finance
 Attn: Cheryl Bowen, Accounts Receivable
 2051 Kaen Road
 Oregon City, OR 97045

## 4. Liaison Responsibility

Liaison from City will be:

Larry Goff, Chief, Lake Oswego Fire Department PO Box 369, Lake Oswego, OR 97034 (503) 697-7410 Igoff@ci.oswego.or.us

Liaison from County will be:

Larry MacDaniels 2051 Kaen Road, Oregon City, OR 97045 (503) 655-8256 larrymac@co.clackamas.or.us

## 5. Other Terms

- A. <u>Compliance with Laws</u>. County and City agree to comply with all applicable local, state, and federal ordinances, statutes, laws and regulations.
- B. <u>No Assignment</u>. This agreement may not be subcontracted, assigned or transferred by either party without the express written consent of the other party.
- C. <u>Entire Agreement; Amendment</u>. This agreement constitutes the entire agreement between the parties, and may be modified only in writing signed by both parties. This agreement may be amended at any time with the written agreement of both parties.
- D. <u>Indemnification and Hold Harmless</u>. Subject to the limitations of liability for public bodies set forth in the Oregon Tort Claims Act, ORS 30.260 to 30.300, and

the Oregon Constitution, each party agrees to hold harmless, defend, and indemnify each other, including its officers, agents, and employees, against all claims, demands, actions and suits (including all attorney fees and costs) arising from the indemnitor's performance of this Agreement where the loss or claim is attributable to the negligent acts or omissions of that party.

- E. <u>Notice of Litigation</u>. Each party shall give the other immediate written notice of any action or suit filed or any claim made against that party that may result in litigation in any way related to this Agreement.
- F. <u>Insurance</u>. Each party agrees to maintain insurance levels or self-insurance in accordance with ORS 30.282, for the duration of this Agreement at levels necessary to protect against public body liability as specified in ORS 30.269 through 30.274.

## 6. Term of Agreement

A. The term of this agreement is a period beginning when it becomes effective and ending one year later. City may elect to renew this Agreement upon the same terms and conditions for additional one-year periods. Renewal shall occur upon written notice to County not sooner than 120 days nor later than 60 days prior to the completion date stated above, and the same date of each year thereafter for which the Agreement is renewed.

## 7. Termination

- A. This agreement may be terminated by either party upon 30 days written notice.
- B. This agreement may be terminated at any time for nonperformance of any material term of this agreement.
- C. This agreement may be terminated at any time by mutual agreement of the County and the City.
- //

//

// (Signature block on next page)

## CLACKAMAS COUNTY BOARD OF COUNTY COMMISSIONERS

## CITY OF LAKE OSWEGO

Chair	Scott Lazenby, City Manager
Attest: Recording Secretary	
Date:	Date:
APPROVED AS TO FORM	APPROVED AS TO FORM
County Counsel	City Attorney

S:\Admin\CONTRACTS\PUBLIC HEALTH\Revenue\City of Lake Oswego\H3SPHCityofLakeOswego7816.docx



July 7, 2016

Board of County Commissioner Clackamas County

Members of the Board:

## Approval of Amendment #3 to the Intergovernmental Agreement with <u>Multnomah County, for a Public Health Officer</u>

Purpose/Outcomes	Amendment #3 extends the term of the agreement one year and adds
Fulpose/Outcomes	
	funding. The Health Officer provides health and medical consultation
	and leadership services and will serve as the Health Officer of record for
	Clackamas County.
Dollar Amount and	Amendment #3 adds \$169,426 for a new contract total of \$504,226
Fiscal Impact	
Funding Source	Local Public Health Authority grant funds through the Oregon Health
	Authority. No County General Funds are involved.
Duration	Effective July 01, 2016 and terminates on June 30, 2017
Strategic Plan	1. Efficient and effective Services
Alignment	2. Build a strong infrastructure
Previous Board	The Original contract was approved by the BCC. June 26, 2014 Agenda
Action	item 062614-A15, Amendment #1 July 9, 2015 Agenda item 070915-A7
Contact Person	Dawn Emerick, Public Health Director – 503-655-8479
Contract No.	6836 -3

## BACKGROUND:

The Clackamas County Public Health Division (CCPHD) of the Health, Housing & Human Services Department requests the approval of Amendment #3 to the Intergovernmental Agreement with Multhomah County. The Health Officer provides health and medical consultation and leadership services and will serve as the Health Officer of record for Clackamas County.

Amendment #1: added \$165,300, extended the term through June 30, 2016 and was approved by the BCC on July 9, 2015.

Amendment #2 removed language from the monthly payment scale and reduced the number of FTE's. This did not require BCC approval.

Amendment #3 adds 169,426. bringing the contract total to \$504,226. Amendment #3 extends the term of the agreement through June 30, 2017. Amendment #3 is effective July 01, 2016 through June 30, 2017. Amendment #3 was reviewed by County Counsel on June 28, 2016. This Amendment is retro-active due to late receipt from Multnomah County.

## **RECOMMENDATION:**

Staff recommends the Board approval of this amendment and authorizes Richard Swift, H3S Director to sign on behalf of Clackamas County.

Respectfully submitted,

Richard Swift, Director Health, Housing, and Human Services

## MULTNOMAH COUNTY INTERGOVERNMENTAL AGREEMENT AMENDMENT

(Amendment to change Contract provisions during contract term)

## Contract Number 201403 Amendment: 3

This is an amendment to Multnomah County's Contract referenced above effective July 1, 2016 between Multnomah County, Oregon, hereinafter referred to as County, and Clackamas County, hereinafter referred to as Contractor.

The parties agree the following changes are made to Contract number 201403:

- 1. The renewal term of this contract is July 1, 2016 to June 30, 2017.
- 2. Any funding adjustment of more than ten percent (10%) between major budget categories (Personnel and Internal Services) shall require the Contractor's prior written approval. Invoices will include a breakdown of current and year-to-date totals for individual personnel and each overall major budget category.
- 3. Attachment B Fiscal Year 2017 Budget is added to this contract.
- 4. \$169,426 in funding is added to this contract for the above referenced renewal term.
- 5. All other terms and conditions of the contract shall remain the same.

#### **MULTNOMAH COUNTY, OREGON:**

#### CONTRACTOR:

County Chair or Designee:	N/A	Signature:	
Date:	N/A	Print Name:	
Dept Director or Designee:		Title:	
Date:		Date:	
REVIEWED:		-	
JENNY M. MADKOUR COUNTY ATTORNEY FOR I By	MULTNOMAH COUNTY	Approved as to form	
Assistant County Attorney		by:	
Date:		Date:	



## Multnomah County Health Department Tri-County Health Officer Program: Clackamas County 201403 Attachment B: Fiscal Year 2017 Budget

					FY 2017 Executive
WBS	WBS Title	Category	Cost Element	Line item	Budget
4CA97-02-1	Clackamas	1-Personnel	60000	Permanent	\$101,407
	County		60130	Salary Related Expns	\$32,156
			60140	Insurance Benefits	\$17,575
		1-Personnel Total			\$151,138
		3-Materials & Services	60180	Printing	\$0
			60220	Repairs and Maint	\$0
			60230	Postage	\$0
			60240	Supplies	\$0
			60260	Travel & Training	\$0
			60270	Local Travel/Mileage	\$0
		<b>3-Materials &amp; Services</b>	Total		\$0
		4-Internal Services	60350	Central Indirect	\$3,990
			60355	Dept Indirect	\$14,298
			60370	Intl Svc Telephone	\$0
			60410	Intl Svc Motor Pool	\$0
			60430	Intl Svc Bldg Mgmt	\$0
		<b>4-Internal Services Tota</b>	al		\$18,288
Grand Total					\$169,426



July 7, 2016

Board of County Commissioners Clackamas County

Members of the Board:

## Approval of a Renewal Grant Agreement with the U.S. Department of Housing and Urban Development, for the Continuum of Care, <u>Homeless Management Information Project</u>

Purpose/Outcomes	Renewal of a federal grant to support homeless programs and
1 dipose/outcomes	services in Clackamas County
Dollar Amount and	Total CoC funds for this grant agreement is \$70,862. No County
Fiscal Impact	General Funds are involved.
Funding Source	The fund source is the HUD Continuum of Care Grant award from the
	US Department of Housing and Urban Development (HUD). No
	County General Funds are involved. Match source is HUD
	Emergency Solutions Grant.
Duration	The term of this grant agreement is 1 year, beginning July 1, 2016
	and ending June 30, 2017.
Previous Board	The Board gave approval to apply for these funds on June 11, 2015.
Action/Review	
Strategic Plan	Provide sustainable and affordable housing.
Alignment	Ensure safe, healthy and secure communities.
Contact Person	Abby Ahern, Housing & Community Development Division, 503-650-
	5663.
Contract No.	H3S 7835

## BACKGROUND:

The Housing & Community Development Division (HCD) of the Department of Health, Housing and Human Services requests the approval of a Renewal Grant Agreement with the U.S. Department of Housing and Urban Development for the Continuum of Care (CoC) Homeless Management Information project. The CoC is a group of individuals and organizations with the common purpose of planning and implementing a housing and services system for people who are homeless. CoC members identify and advocate for the needs of people who are homeless and develop short and long term plans to mitigate homelessness in Clackamas County.

HUD requires that Clackamas County maintains a Homeless Management Information System (HMIS) to collect data on all the homeless programs and services in Clackamas County. Community Development staff manage this database for our partner agencies. We use this system to track outcomes, analyze data, and ensure appropriate use of HUD CoC grant funds by our community partners.

## **RECOMMENDATION:**

Staff recommends Board approval of this grant agreement and authorization for Richard Swift, H3S Director to sign on behalf of Clackamas County.

Respectfully submitted,

Richard Swift, Director Health, Housing and Human Services

## Tax ID No.: 93-6002286 CoC Program Grant Number: OR0099L0E071508 (HMIS) Effective Date: June 17, 2016 DUNS No.: 096992656

## CONTINUUM OF CARE PROGRAM Grant Agreement

This Grant Agreement ("this Agreement") is made by and between the United States Department of Housing and Urban Development ("HUD") and <u>Clackamas Dept. Health, Housing & Human Srvs</u> (the "Recipient").

This Agreement is governed by title IV of the McKinney-Vento Homeless Assistance Act 42 U.S.C. 11301 *et seq.* (the "Act") and the Continuum of Care Program rule (the "Rule").

The terms "Grant " or "Grant Funds" mean the funds that are provided under this Agreement. The term "Application" means the application submissions on the basis of which the Grant was approved by HUD, including the certifications, assurances, and any information or documentation required to meet any grant award condition. All other terms shall have the meanings given in the Rule.

The Application is incorporated herein as part of this Agreement, except that only those projects listed, and only in the amounts listed on a Scope of Work exhibit, are funded by this Agreement. In the event of any conflict between any application provision and any provision contained in this Agreement, this Agreement shall control.

Exhibit 1, the FY2015 Scope of Work, is attached hereto and made a part hereof. If in the future appropriations are made available for Continuum of Care grants; if Recipient applies under a Notice of Funds Availability published by HUD; and, if pursuant to the selection criteria in the Notice of Funds Availability, HUD selects Recipient and one or more projects listed on Exhibit 1 for renewal, then additional Scope of Work exhibits may be attached to this Agreement. Those additional exhibits, when attached, will also become a part hereof.

The effective date of the Agreement shall be the date of execution by HUD and it is the date use of funds under this Agreement may begin. Each project will have a performance period that will be listed on the Scope of Work exhibit(s) to this Agreement. For renewal projects, the period of performance shall begin at the end of the Recipient's final operating year for the project being renewed and eligible costs incurred for a project between the end of Recipient's final operating year under the grant being renewed and the execution of this Agreement may be paid with funds from the first operating year of this Agreement. For each new project funded under this Agreement, Recipient and HUD will set an operating start date in *e*LOCCS, which will be used to track expenditures, to establish the project performance period and to determine when a project is eligible for renewal. Recipient hereby authorizes HUD to insert the project performance period for new projects into the exhibit without Recipient signature, after the operating start date is established in *e*LOCCS.

This Agreement shall remain in effect until termination either: 1) by agreement of the parties; 2) by HUD alone, acting under the authority of 24 CFR 578.107; 3) upon expiration of the final performance period for all projects funded under this Agreement; or 4) upon the expiration of the period of availability of funds for all projects funded under this Agreement.

Recipient agrees:

1. To ensure the operation of the project(s) listed on the Scope of Work in accordance with the

provisions of the Act and all requirements of the Rule;

- 2. To monitor and report the progress of the project(s) to the Continuum of Care and HUD;
- 3. To ensure, to the maximum extent practicable, that individuals and families experiencing homelessness are involved, through employment, provision of volunteer services, or otherwise, in constructing, rehabilitating, maintaining, and operating facilities for the project and in providing supportive services for the project;
- 4. To require certification from all subrecipients that:
  - a. Subrecipients will maintain the confidentiality of records pertaining to any individual or family that was provided family violence prevention or treatment services through the project;
  - b. The address or location of any family violence project assisted with grant funds will not be made public, except with written authorization of the person responsible for the operation of such project;
  - c. Subrecipients will establish policies and practices that are consistent with, and do not restrict, the exercise of rights provided by subtitle B of title VII of the Act and other laws relating to the provision of educational and related services to individuals and families experiencing homelessness;
  - d. In the case of projects that provide housing or services to families, that subrecipients will designate a staff person to be responsible for ensuring that children being served in the program are enrolled in school and connected to appropriate services in the community, including early childhood programs such as Head Start, part C of the Individuals with Disabilities Education Act, and programs authorized under subtitle B of title VII of the Act;
  - e. The subrecipient, it officers, and employees are not debarred or suspended from doing business with the Federal Government; and
  - f. Subrecipients will provide information, such as data and reports, as required by HUD; and
- 5. To establish such fiscal control and accounting procedures as may be necessary to assure the proper disbursal of, and accounting for grant funds in order to ensure that all financial transactions are conducted, and records maintained in accordance with generally accepted accounting principles, if the Recipient is a Unified Funding Agency;
- 6. To monitor subrecipient match and report on match to HUD;
- 7. To take the educational needs of children into account when families are placed in housing and will, to the maximum extent practicable, place families with children as close as possible to their school of origin so as not to disrupt such children's education;
- 8. To monitor subrecipients at least annually;
- 9. To use the centralized or coordinated assessment system established by the Continuum of Care as required by the Rule. A victim service provider may choose not to use the Continuum of Care's centralized or coordinated assessment system, provided that victim service providers in the area use a centralized or coordinated assessment system that meets HUD's minimum requirements and the victim service provider uses that system instead;

- 10. To follow the written standards for providing Continuum of Care assistance developed by the Continuum of Care, including those required by the Rule;
- 11. Enter into subrecipient agreements requiring subrecipients to operate the project(s) in accordance with the provisions of this Act and all requirements of the Rule; and

12. To comply with such other terms and conditions as HUD may have established in the applicable Notice of Funds Availability.

HUD notifications to the Recipient shall be to the address of the Recipient as stated in the Application, unless HUD is otherwise advised in writing. Recipient notifications to HUD shall be to the HUD Field Office executing the Agreement. No right, benefit, or advantage of the Recipient hereunder may be assigned without prior written approval of HUD.

The Agreement constitutes the entire agreement between the parties hereto, and may be amended only in writing executed by HUD and the Recipient.

By signing below, Recipients that are states and units of local government certify that they are following a current HUD approved CHAS (Consolidated Plan).

This agreement is hereby executed on behalf of the parties as follows:

## UNITED STATES OF AMERICA, Secretary of Housing and Urban Development

BY:

(Signature)

Doug Carlson. Director, Community Planning and Development (Typed Name and Title)

3

è.

<u>June 17, 2016</u> (Date)

#### RECIPIENT

<u>Clackamas County Dept. Health. Housing & Human Srvs</u> (Name of Organization)

BY:

(Signature of Authorized Official)

(Typed Name and Title of Authorized Official)

(Date)

Tax ID No.: 93-6002286 CoC Program Grant Number: OR0099L0E071508 (HMIS) Effective Date: June 17, 2016 DUNS No.: 096992656

## EXHIBIT 1 SCOPE OF WORK for FY2015 COMPETITION

- The project listed on this Scope of Work is governed by the Continuum of Care program Interim Rule attached hereto and made a part hereof as Exhibit 1a. Upon publication for effect of a Final Rule for the Continuum of Care program, the Final Rule will govern this Agreement instead of the Interim Rule. The project listed on this Exhibit at 4 below, is also subject to the terms of the Notice of Funds Availability for the fiscal year listed above.
- 2. The Continuum that designated Recipient to apply for grant funds has not been designated a high performing community by HUD for the applicable fiscal year.
- Recipient is not the only Recipient for the Continuum of Care. HUD's total funding obligation for this grant and project is \$70,862.00 allocated between budget line items, as indicated in 4. below.
- 4. HUD agrees, subject to the terms of this Agreement, to provide the Grant funds for the project application listed below in the amount specified below to be used during the performance period established below. However, no funds for new projects may be drawn down by Recipient until HUD has approved site control pursuant to the Rule and no funds for renewal projects may be drawn down by Recipient before the end date of the project's final operating year under the grant that has been renewed.

#### Project No. OR0099

Performance Period: July 1, 2016 – June 30, 2017

The obligation for this project shall be allocated between budget line items as follows:

a. Continuum of Care planning activities	\$
b. UFA costs	\$
c. Acquisition	\$
d. Rehabilitation	\$
e. New construction	\$
f. Leasing	\$
g. Rental assistance	\$

(of which \$\_\_\_\_\_ is for short-term and medium-term rental assistance for persons at risk of homelessness)

h. Supportive services	\$
i. Operating costs	\$
j. Homeless Management Information System	\$ <u>66,372</u>
k. Administrative costs	\$4,490
I. Relocation costs	\$
m. Housing relocation and stabilization services	\$

In accordance with the Rule, Recipient is prohibited from moving more than 10% from one budget line item in a project's approved budget to another without written amendment to this Agreement.

- 5. If grant funds will be used for payment of indirect costs, pursuant to 2 CFR 200, Subpart E Cost Principles, the Recipient is authorized to insert the Recipient's federally recognized indirect cost rates (including if the de minimis rate is charged per 2 CFR §200.414) on the attached Federally Recognized Indirect Cost Rates Schedule, which Schedule shall be incorporated herein and made a part of the Agreement. No indirect costs may be charged to the grant by the Recipient if their federally recognized cost rate is not listed on the Schedule. Do not include indirect cost rates for Subrecipients, however, Subrecipients may not charge indirect costs to the grant if they do not also have a federally recognized indirect cost rate.
- 6. The project has/has not been awarded project-based rental assistance for a term of fifteen (15) years. Funding is provided under this Scope of Work for the performance period stated in paragraph 4. Additional funding is subject to the availability of annual appropriations.
- 7. Program income earned during the grant term shall be retained and may either be 1) added to funds committed to the project by HUD and the recipient and used for eligible activities; or 2) used as match.

# DRAFT

# Approval of Previous Business Meeting Minutes: June 16, 2016

(draft minutes attached)

## **BOARD OF COUNTY COMMISSIONERS BUSINESS MEETING MINUTES**

A complete video copy and packet including staff reports of this meeting can be viewed at http://www.clackamas.us/bcc/business.html

Thursday, June 16, 2016 – 6:00 PM

Public Services Building

2051 Kaen Rd., Oregon City, OR 97045

PRESENT: Commissioner Tootie Smith Commissioner Jim Bernard Commissioner Paul Savas Commissioner Martha Schrader EXECUSED: Commissioner John Ludlow, Chair

## CALL TO ORDER

Roll Call

Chair Ludlow is out of the office today, Commissioner Tootie Smith will serve as Chair for this meeting.

Pledge of Allegiance

## I. CITIZEN COMMUNICATION

http://www.clackamas.us/bcc/business.html

1. Jerry Myra, Oregon City – very concerned regarding a marijuana grow next door to him.

~Board Discussion including Don Krupp and Jennifer Hughes~

- 2. Mack Woods, Canby support of Veterans.
- 3. Dena Kronburg Swanson, Tigard Gave history of Kronberg Park.
- 4. Jose Hernandez, Milwaukie tax exempt for Veterans.
- 5. Garry McCadems, stated the Oregon Dept. of Veterans Affairs has agents to assist Veterans, Mon. Friday 8 AM to 5 PM.
- 6. Dr. Kendal Evans, Oregon City opposes marijuana grow regulations from the County.
- 7. Richard Langdon, Portland comments regarding Commissioner Bernard.
- 8. Tina Kennedy, Milwaukie transitional housing facility for homeless veterans. Tina was speaking on behalf of 11 other folks in the audience, including Paul Edgar.
- 9. Les Poole, Gladstone Robert Kronburg Park.
- 10. Skip Ormsby, Birds Hill CPO requested a sight tour of the Terwilliger Rd. project with ODOT, City of Portland, City of Lake Oswego, Clackamas County and Citizens.

~Board Discussion~

## II. PUBLIC HEARING

1. **Resolution No. 2016-44** Approving Clackamas County Supplemental Budget (Greater than 10% and Budget Reduction) for Fiscal Year 2015-2016

Diane Padilla, Finance presented the staff report.

~Board Discussion~

Chair Ludlow opened the public hearing and asked if anyone wishes to speak, seeing none he closed the public hearing and asked for a motion.

## MOTION:

Chair Smith:

Commissioner Bernard:	I move we approve the Resolution for a Clackamas County Supplemental Budget (Greater than 10% and Budget Reduction) for Fiscal Year 2015-2016.
	,
Commissioner Savas:	Second.
Clerk calls the poll.	
Commissioner Bernard:	Aye.
Commissioner Schrader:	Aye.
Commissioner Savas:	Aye.

Aye - the motion passes 4-0.

## III. PUBLIC DISCUSSION ITEM

1. **Resolution No. 2016-45** Certifying Election Results Authorizing General Obligation Bonds to Replace Obsolete Emergency Radio Communications

Laurel Butman, County Administration and Chris Storey, County Counsel presented the staff report.

~Board Discussion~

Chair Ludlow announced this is a discussion item and asked if anyone wishes to speak, seeing none he asked for a motion.

#### MOTION:

Reduction) for Fiscal Year 2015-2016.
Commissioner Schrader: Second. <i>~Board Discussion~</i> Clerk calls the poll.
Commissioner Schrader: Aye.
Commissioner Savas: Aye.
Commissioner Bernard: Aye.
Chair Smith: Aye – the motion passes 4-0.

## IV. CONSENT AGENDA

Chair Ludlow asked the Clerk to read the consent agenda by title, he then asked for a motion.

## **MOTION:**

I move we approve the consent agenda. Second.
Aye.
Aye.
Aye.
Aye – the motion passes 4-0.

## A. Department of Transportation & Development

1. **Resolution No. 2016-46** Directing the Disposition of Clackamas County Development Agency Surplus Property at Public Oral Auction & Acceptance of Clackamas County Development Agency Surplus Property

## B. Finance Department

- 1. **Resolution No. 2016-47** Approving Clackamas County for Budgeting of New Specific Purpose Revenue for Fiscal Year 2015-2016
- 2. **Resolution No. 2016-48** Approving Clackamas County Supplemental Budget (Less than 10 %) for Fiscal Year 2015-2016
- 3. **Resolution No. 2016-49** Approving Clackamas County for Transfer of Appropriations for Fiscal Year 2015-2016

## C. Elected Officials

1. Approval of Previous Business Meeting Minutes – *BCC* 

## Page 3 - Business Meeting Minutes - June 16, 2016

## D. Department of Communications (C-COM)

1. Approval of an Amended Intergovernmental Agreement with Washington County Consolidated Communications Agency (WCCCA), the City of Lake Oswego 911, (LOCOM), and Columbia 911 Communications District (C911CD)

#### E. <u>County Counsel</u>

- 1. Approval of the Designation of Newspaper for 2016 Property Tax Foreclosure Publication
- 2. Approval of Lease Amendment No. 2 Between Clackamas County and OSU Extension in Clackamas County

#### F. Business & Community Services

1. Approval of a Modification of Grant Agreement 13-SA-11060600-013 with the USDA Forest Service, Mt. Hood National Forest for the Dump Stoppers Program

#### V. DEVELOPMENT AGENCY

- 1. **Resolution No. 2016-50** Authorizing the Disposition of Clackamas County Development Agency Surplus Property
- 2. Resolution No. 2016-51 Authorizing the Disposition of Clackamas County Development Agency Surplus at Public Oral Auction
- 3. Approval of a Contract with Moore Excavation, Inc. for the Otty Street Realignment Project – *Procurement*

## VI. COUNTY ADMINISTRATOR UPDATE

http://www.clackamas.us/bcc/business.html

## VII. COMMISSIONERS COMMUNICATION

http://www.clackamas.us/bcc/business.html

## **MEETING ADJOURNED -**

NOTE: Regularly scheduled Business Meetings are televised and broadcast on the Clackamas County Government Channel. These programs are also accessible through the County's Internet site. DVD copies of regularly scheduled BCC Thursday Business Meetings are available for checkout at the Clackamas County Library in Oak Grove. You may also order copies from any library in Clackamas County or the Clackamas County Government Channel. www.clackamas.us/bcc/business.html



Gregory L. Geist Director

July 7, 2016

Board of County Commissioners Clackamas County

Members of the Board:

## Approval of an Order Authorizing Refunding Revenue Bonds to Reduce Debt Service Expense for Clackamas County Service District No. 1

Purpose/Outcomes	Approval will allow Clackamas County Service District No. 1 ("CCSD #1") to refund outstanding bonds to achieve annual debt service savings and to unencumber certain cash reserves currently required to be maintained as debt service reserves.
Dollar Amount and	Costs for professional services related to the refunding will be paid from the
Fiscal Impact	proceeds of the refunding. The fiscal impact will be reduced annual debt
	service costs for CCSD#1. The current estimate of savings is in the range of
	\$400,000 to \$800,000 annually.
Funding Source	Bond proceeds for CCSD #1. No General Funds are impacted.
Duration	The refunding is expected to be completed by the end of August 2016.
Previous Board	Approval Amending Order No. 2008-108 to Remove Certain Conditions –
Action/Review	092409 2009-99
	Authorizing Conditional Rate Increases to Support the Issuance of Revenue Obligations to Finance Wastewater Treatment and Conveyance Infrastructure – 092409 2009-100
Strategic Plan Alignment	<ol> <li>Supports WES' Budget and Financial Planning Program, specifically the result: By the end of Fiscal Year 2017/2018, WES' priorities and policy recommendations will reflect optimum economies of scale, defined as lowest rate per user per district to achieve the 20 year Comprehensive Plan.</li> <li>Supports the County strategic goal of building public trust through good government.</li> </ol>
Contact Person	Doug Waugh 503.742.4564 dougwau@clackamas.us
Contract No.	N/A

## BACKGROUND:

Interest rates on all borrowings typically done by government are at or near all-time lows. Consequently, there is an opportunity to create significant annual savings on the bonds issued by CCSD #1 in 2009 and 2010. In order to realize these savings, CCSD #1 needs to refund the borrowings done in 2009 and 2010 with new borrowings. These new borrowings are authorized by the Resolution and Order and the Master Sewer Revenue Bond Declaration attached to this staff report. These new borrowings in turn will partially reduce the need for future adjustments to the CCSD #1's monthly sewer fee due to the annual savings created. The RiverHealth Advisory Committee approved this refunding at their meeting on May 23, 2016.

This Resolution and Order and the Master Sewer Revenue Bond Declaration have been reviewed and approved by County Counsel.

## **RECOMMENDATION:**

CCSD #1 staff recommends the Board of County Commissioners of Clackamas County, acting as the governing body of Clackamas County Service District No. 1, approve the Order Authorizing Refunding Revenue Bonds to Reduce Debt Service Expense for Clackamas County Service District No. 1.

Respectfully submitted,

Greg Geist, Director Water Environment Services

## BEFORE THE BOARD OF COUNTY COMMISSIONERS OF CLACKAMAS COUNTY, STATE OF OREGON

)

)

)

)

In the matter of authorizing refunding revenue bonds to reduce debt service expense for Clackamas County Service District No. 1 and providing for related matters.

Order No. (Page 1 of 2)

This matter coming before the Board of County Commissioners of Clackamas County, Oregon, (the "Board") acting as the governing body of Clackamas County Service District No. 1 (the "District") and it appearing to the Board that the District is authorized by Oregon Revised Statutes Sections 287A.360 to 287A.375 to issue refunding bonds to refund outstanding obligations of the District; and

It appearing that the District may obtain debt service savings by refunding some of its outstanding obligations; and

It appearing that the District's outstanding bonds were issued under District Order No. 94-1085, as amended, which specifies the District's obligations to its bond owners; and

It appearing that, with the consent of the purchasers of the refunding bonds, the District may amend that order and replace it with a Master Sewer Revenue Bond Declaration substantially in the form attached hereto as <u>Exhibit A</u> (the "Master Declaration") which will give the district greater flexibility;

NOW THEREFORE, IT IS HEREBY RESOLVED AND ORDERED that the District Administrator or the District Director (each of whom is referred to as a "District Official") are each hereby authorized, on behalf of the District and without further action by the Board, to:

1. Select the District's outstanding sewer revenue bonds and loans that may be refunded to obtain debt service savings (the "Refundable Obligations").

2. Negotiate, execute and deliver the Master Declaration, with any changes that a District Official determines are advantageous to the District and/or consistent with the purposes of this resolution and order.

3. Determine their final terms and issue revenue refunding bonds (the "Refunding Bonds") to refund the Refundable Obligations pursuant to the Master Sewer Revenue Bond Declaration. The Refunding Bonds may be issued in one or more series. The aggregate principal amount of the Refunding Bonds shall not exceed the amount the District Official estimates will be required to refund the Refundable Obligations, and to pay costs related to the Refunding Bonds.

4. Determine whether the interest payable on each series of Refunding Bonds will be taxable (includable in gross income under the Internal Revenue Code of 1986, as amended (the "Code")), or tax-exempt (excludable from gross income under the Code).

5. Covenant for the benefit of and as an investment inducement to the owners of tax-exempt Refunding Bonds to comply with all provisions of the Code which are required for the interest on

that series of Refunding Bonds to be excluded from gross income for federal income tax purposes.

6. Deem final and authorize the distribution of a preliminary official statement for each series of Obligations, authorize the preparation and distribution of a final official statement or other disclosure document for each series of Refunding Bonds, and enter into agreements to provide continuing disclosure for owners of each series of Refunding Bonds.

7. Undertake to provide continuing disclosure for each series of Refunding Bonds in accordance with Rule 15c2-12 of the United States Securities and Exchange Commission.

8. Apply for ratings for each series of Refunding Bonds, determine whether to purchase municipal bond insurance, reserve credit facilities or obtain other forms of credit enhancement for each series of Refunding Bonds, enter into agreements with the providers of credit enhancement, and execute and deliver related documents.

9. Engage the services of escrow agents or trustees and any other professionals whose services are desirable for the refunding.

10. Solicit competitive bids for the purchase of any series of the Refunding Bonds and award their sale to the bidder offering the most favorable terms to the District.

11. Authorize submission of one or more advance refunding plans for the Refundable Obligations to the State Treasurer, call the Refundable Obligations for redemption, contribute amounts in existing debt service funds and debt service reserves to the refunding, and defease the Refundable Obligations.

12. Execute and deliver any other certificates, agreements or documents and take any other actions which the District Official determines are desirable to carry out this resolution and order.

DATED this 7<sup>th</sup> day of July, 2016.

Chair

**Recording Secretary** 

Exhibit A

**MASTER SEWER REVENUE BOND DECLARATION** 

## **CLACKAMAS COUNTY SERVICE DISTRICT NO. 1**

**CLACKAMAS COUNTY, OREGON** 

\_\_\_\_\_, 2016

## **Table of Contents**

Pag	<u>e</u>
Section 1. Master Declaration1	
Section 2. Recitals	
Section 3. Definitions	
Section 4. Deposit, Pledge and Use of Gross Revenues	
Section 5. Bond Funds and Accounts10	
Section 6. Rate Covenant14	
Section 7. Parity Obligations16	
Section 8. Subordinate Obligations	
Section 9. Separate Utility System	
Section 10. General Covenants	
Section 11. Events of Default and Remedies	
Section 12. Bondowners Meetings	
Section 13. Amendment of Master Declaration	
Section 14. Defeasance	
Section 15. BEO System	
Section 16. Redemption of Bonds	
Section 17. Authentication, Registration and Transfer	
Section 18. The Series 2016 Bonds	
Section 19. Tax-Exempt Status	

#### MASTER SEWER REVENUE BOND DECLARATION

#### Section 1. Master Declaration

THIS MASTER SEWER REVENUE BOND DECLARATION is executed as of \_\_\_\_\_\_, 2016, by the Authorized Officer of Clackamas County Service District No. 1, Clackamas County, Oregon.

#### Section 2. Recitals

**2.1.** The Board of County Commissioners of Clackamas County, Oregon, acting as the governing body of the District, adopted Order No. 94 1085 on September 15, 1994, which established the terms under which the District could issue obligations secured by the Net Revenues of the Sewer System. The District's outstanding Series 2002B Bonds, Series 2009A Obligations, Series 2009B Obligations and Series 2010 Obligations were issued under Order No. 94 1085, as amended (the "Order").

**2.2.** Section 12.2 of the Order allows the District to amend the Order with the consent of not less than 51% in aggregate principal amount of the Bonds that are outstanding at the time of the amendment, so long as the amendment does not: (i) extend the maturity of any Bond, reduce the rate of interest upon any Bond, extend the time of payment of interest on any Bond, reduce the amount of principal payable on any Bond, or reduce any premium payable on any Bond; or (ii) reduce the percent of Bondowners required to approve amendatory orders.

**2.3.** The District executes this Master Declaration to replace the Order and amend, modernize and restate its terms. The amendments to the Order that are made by this Master Declaration do not: (i) extend the maturity of any Bond, reduce the rate of interest upon any Bond, extend the time of payment of interest on any Bond, reduce the amount of principal payable on any Bond, or reduce any premium payable on any Bond; or (ii) reduce the percent of Bondowners required to approve amendatory Master Declarations.

**2.4.** Concurrently with the execution of this Master Declaration the District is issuing its Sewer Revenue and Refunding Bonds, Series 2016, and refunding and defeasing certain of the District's previously issued Bonds. By accepting delivery of the Series 2016 Bonds their Owners will be deemed to have consented to the amendments to the Order that are made by this Master Declaration. Because the aggregate principal amount of the Series 2016 Bonds exceeds 51% in aggregate principal amount of all the Bonds that will be outstanding when the Series 2016 Bonds are issued, and the owners of the Series 2016 Bonds have consented to the amendments to the Order that are made by this Master Declaration, the Owners of more than 51% in aggregate principal amount of all the Bonds have consented to the amendments to the Order that are made by this Master Declaration, the Owners of more than 51% in aggregate principal amount of all the Bonds have consented to the amendments to the Order that are made by this Master Declaration, the Owners of more than 51% in aggregate principal amount of all the Bonds have consented to the amendments to the Order that are made by this Master Declaration, the Owners of more than 51% in aggregate principal amount of all the Bonds have consented to the amendments to the Order that are made by this Master Declaration, the Owners of more than 51% in aggregate principal amount of all the Bonds have consented to the amendments to the Order that are made by this Master Declaration.

**2.5.** This Master Declaration describes the covenants of the District that apply to the District's outstanding Bonds, including the Series 2016 Bonds, and to all Bonds of the District that are issued in the future.

Capitalized terms used in this Master Declaration shall have the following meanings unless the context clearly requires use of a different meaning:

**3.1.** "Annual Debt Service" means the amount of principal and interest on Outstanding Bonds which is required to be paid in a Fiscal Year, calculated as follows:

(A) interest which is to be paid from Bond Proceeds shall be subtracted;

(B) District Payments to be made in the Fiscal Year under a Parity Derivative Product shall increase Annual Debt Service, and Reciprocal Payments to be received in the Fiscal Year under a Parity Derivative Product shall reduce Annual Debt Service;

(C) Bonds which are subject to scheduled, noncontingent redemption or tender shall be deemed to mature on the dates and in the amounts which are subject to mandatory redemption or tender, and only the amount scheduled to be outstanding on the final maturity date shall be treated as maturing on that date;

(D) Bonds which are subject to contingent redemption or tender shall be treated as maturing on their stated maturity dates;

(E) Interest subsidies shall be subtracted from the interest due on Interest Subsidy Bonds as provided in Section 6.4; and,

(F) Variable Rate Obligations shall be deemed to bear interest from each respective date of computation until their scheduled maturity date at their Estimated Average Interest Rate, calculated as of the date of computation.

**3.2.** "Audit" means the audit, if any, required by ORS 297.425, as it may be amended from time to time.

**3.3. "Auditor"** means a person authorized by the State Board of Accountancy to conduct municipal audits pursuant to ORS 297.670.

**3.4. "Authorized Officer"** means the District Administrator or the District Director.

**3.5. "Base Period"** means any twelve consecutive months selected by the District out of the most recent twenty-four months preceding the delivery of a Series of Parity Obligations.

**3.6. "Board"** means the Board of County Commissioners of Clackamas County, acting as the governing body of the District, or its successors.

**3.7. "Bond Counsel"** means Hawkins Delafield & Wood LLP or another law firm having knowledge and expertise in the field of municipal law and which offers opinions on municipal bonds which are generally accepted by purchasers of municipal bonds.

3.8. "Bondowner" or "Owner" means a registered owner of a Bond.

**3.9. "Bondowners Committee"** means that committee described in Section 13.

**3.10. "Bonds"** means the outstanding Series 2002B Bonds, Series 2009A Obligations, Series 2009B Obligations, Series 2010 Obligations, Series 2016 Bonds and any Parity Obligations.

**3.11. "BEO" or "Book-Entry-Only System"** means a system for clearance and settlement of securities transactions through electronic book-entry changes, which eliminates the need for physical movement of securities.

**3.12. "Business Day"** means any day except a Saturday, a Sunday, a legal holiday a day on which the offices of banks in Oregon or New York are authorized or required by law or executive order to remain closed, or a day on which the New York Stock Exchange is closed.

**3.13. "Code"** means the Internal Revenue Code of 1986, as amended, including the rules and regulations promulgated thereunder.

**3.14. "Construction Account"** means the Sewer Construction Account in the Sewer Enterprise Fund, which the District has created to hold proceeds of Bonds and other revenues related to capital improvements.

**3.15. "Credit Facility"** means a letter of credit, a municipal bond insurance policy, standby bond purchase agreement, or other credit enhancement device which is obtained by the District, which provides for payment in full of Bonds, and which is issued or provided by a Credit Provider.

**3.16. "Credit Provider"** means the person or entity that is: (i) obligated to make or guarantee payments under a Credit Facility; and (ii) whose long-term debt obligations or claims-paying ability (as appropriate) are rated, at the time the Credit Facility is issued, in one of the two highest rating categories, determined without regard to pluses, minuses or other degrees, by a Rating Agency which has issued a rating on Outstanding Bonds. Under rating systems in effect on the date of this Master Declaration, a rating in one of the three highest rating categories by a Rating Agency would be a rating in the "AA" category or better.

**3.17.** "**Depository**" or "**DTC**" means The Depository Trust Company or any other qualified securities depository designated by the District as its successor.

**3.18. "Derivative Product"** means a written contract between the District and a Reciprocal Payor under which the District is obligated to pay the District Payments in exchange for the Reciprocal Payor's obligation to pay Reciprocal Payments; and which provides that the District is not required to fulfill its obligations under the contract if:

(Å) the Reciprocal Payor fails to make any Reciprocal Payment; or

(B) the Reciprocal Payor fails to comply with its financial status covenants.

**3.19. "Direct Obligations"** means direct obligations of the United States, and any obligations the payment of which is fully and unconditionally guaranteed by the United States.

**3.20.** "District" means the Clackamas County Service District No. 1, Clackamas County, Oregon, a municipal corporation of the State of Oregon.

**3.21. "District Payment"** means any scheduled payment required to be made by or on behalf of the District under a Derivative Product which is either fixed in amount or is determined according to a formula set forth in the Derivative Product.

3.22. "Estimated Average Interest Rate" is the rate at which Variable Rate Obligations are assumed to bear interest. Estimated Average Interest Rate shall be assumed to be equal to (1) if those Bonds were Outstanding during the 12 calendar months immediately preceding the date of calculation, an average of the interest rates per annum which were in effect, and (2) if those Bonds were not Outstanding during the 12 calendar months immediately preceding the date of calculation, (A) with respect to Tax Exempt Bonds, an average of the SIFMA Index during the 12 calendar months immediately preceding the date of calculation, (B) with respect to Bonds that are not Tax-Exempt Bonds, an average of an index determined by the District to be comparable to SIFMA for Bonds that are not Tax Exempt Bonds during the 12 calendar months immediately preceding the date of calculation, (C) with respect to Derivative Products with an index-based rate formula, the rate produced by applying that rate formula to an average of such index during the 12 calendar months immediately preceding the date of calculation, or (D) with respect to Derivative Products that do not have an index-based rate, the rate described in (A) above if the related Bonds are Tax-Exempt Bonds or in (B) above if the related Bonds are taxable, all as specified in either, at the election of the District, a certificate of the District Official or a written statement from an investment banking or financial advisory firm.

3.23. "Event of Default" means any event specified in Section 11.2 of this Master Declaration.

**3.24. "First Reserve Subaccount"** means the subaccount of the Revenue Bond Reserve Account that secures the Series 2002B Bonds, Series 2009A Obligations, Series 2009B Obligations, Series 2010 Obligations, Series 2016 Bonds and any subsequent Series of Bonds to which the amounts in the First Reserve Subaccount are pledged and is described in Section 5.4.

**3.25. "First Reserve Subaccount Reserve Requirement"** means an amount equal to the Maximum Annual Debt Service on all Outstanding Bonds that are secured by the First Reserve Subaccount or the amount described in the next sentence. If at the time of issuance of a Series of Bonds secured by the First Reserve Subaccount, the amount required to be added to the First Reserve Subaccount to make the balance in the First Reserve Subaccount equal to Maximum Annual Debt Service on all Outstanding Bonds that are secured by the First Reserve Subaccount exceeds the Tax Maximum calculated with respect to such Series of Bonds, then the First Reserve Requirement in effect immediately prior to the issuance of that Series of Bonds, plus the Tax Maximum calculated with respect to that Series of Bonds. On the date of Closing of the Series 2016 Bonds, the First Reserve Subaccount Reserve Requirement is \$\_\_\_\_\_\_, which is equal to \_\_\_\_\_\_.

**3.26.** "First Reserve Subaccount Valuation Date" means the first Business Day of each Fiscal Year, each date on which amounts are withdrawn from the First Reserve Subaccount, and each Closing date for a Series of Bonds that is secured by the First Reserve Subaccount.

**3.27. "Fiscal Year"** means the fiscal year of the District as established by Oregon Law. At the time this Master Declaration is adopted, the fiscal year of the District begins on July 1 of each calendar year and ends on the next succeeding June 30.

3.28. "Fitch" means Fitch Ratings, its successors and assigns.

**3.29.** "Gross Revenues" means all revenues, fees and charges and other revenues resulting from the operation of the Sewer System, including systems development charges, revenues from product sales and interest earnings on Gross Revenues in the Sewer Enterprise Fund. However, the term "Gross Revenues" does not include:

(A) the interest income or other earnings derived from the investment of the Rebate Fund or any escrow fund established for the defeasance or refunding of outstanding indebtedness of the District;(B) any gifts, grants, donations or other moneys received by the District from any State or Federal District or other person if such moneys are restricted by law or the grantor to uses inconsistent with the payment of Bonds;

(C) the proceeds of any borrowing;

(D) the proceeds of any liability or other insurance (excluding business interruption insurance or other insurance of like nature insuring against the loss of revenues);

(E) the proceeds of any casualty insurance which the District intends to utilize for repair or replacement of the Sewer System;

(F) the proceeds derived from the sales of assets pursuant to Section 10.7 of this Master Declaration;

(G) any ad valorem or other taxes imposed by the District (except charges or payments for Sewer System services which become "taxes" within the meaning of Article XI, Section 11b of the Oregon Constitution only because they are imposed on property);

(H) any income, fees, charges, receipts, profits or other moneys derived by the District from its ownership or operation of any Separate Utility System; and

(I) Any federal interest subsidies the District receives for Interest Subsidy Bonds.

**3.30. "Interest Payment Date"** means any date on which Bond interest is scheduled to be paid, and any date on which Bonds are called for redemption.

**3.31. "Interest Subsidy Bonds**" means Bonds for which the District is eligible to receive federal interest rate subsidies that are similar to the interest subsidies that were available for Build America Bonds.

3.32. "Master Declaration" means this Master Sewer Revenue Bond Declaration.

**3.33. "Maximum Annual Debt Service**" means the greatest Annual Debt Service, calculated on all Bonds which are Outstanding on the date of calculation.

**3.34.** "Moody's" means Moody's Investors Service, a corporation organized and existing under the laws of the State of Delaware, its successors and their assigns.

3.35. "Net Revenues" means the Gross Revenues less the Operating Expenses.

**3.36.** "Operating Expenses" means all costs which are properly treated as expenses of operating and maintaining the Sewer System under generally accepted accounting principles or rules of accounting applicable to municipal enterprises similar to the Sewer System. However, Operating Expenses do not include:

(Å) any rebates or penalties paid from Gross Revenues under Section 148 of the Code;

(B) payments of judgments against the District and payments for the settlement of litigation;

(C) depreciation and amortization of property values or losses, other non-cash expenses and all amounts treated for accounting purposes as payments for capital expenditures;

(D) debt service payments;

(E) the expenses of owning, operating or maintaining any Separate Utility System;

(F) extraordinary non-recurring expenses of the Sewer System; or

(G) franchise fees and similar charges imposed by the District or Clackamas County on the Sewer System or its operations.

3.37. "ORS" means the Oregon Revised Statutes.

**3.38.** "Outstanding" refers to all Bonds authorized and delivered pursuant to this Master Declaration and any Supplemental Declaration except Bonds theretofore canceled or defeased pursuant to Section 14 of this Master Declaration, and Bonds which have matured and not been presented for payment (provided sufficient funds to pay those Bonds have been transferred to the Paying Agent).

**3.39. "Parity Derivative Product"** means a Derivative Product which qualifies as a Parity Obligation in accordance with Section 7.5.

**3.40. "Parity Obligation"** means any obligation payable from the Net Revenues which is issued in accordance with Section 7, and includes any Parity Derivative Product.

3.41. "Payment Date" means a Principal Payment Date or an Interest Payment Date.

**3.42. "Permitted Investments"** means any investments which the District is permitted to make under the laws of the State.

**3.43. "Principal Payment Date"** means any date on which any Bonds are scheduled to be retired, whether by virtue of their maturity or by mandatory sinking fund redemption prior to maturity, and the redemption date of any Bonds which have been called for redemption.

3.44. "Project" means any lawful purpose for which Gross Revenues may be spent.

**3.45. "Qualified Consultant"** means an independent engineer, an independent auditor, an independent municipal advisor, or similar independent professional consultant of recognized standing and having experience and expertise in the area for which such person or firm is retained by the District for purposes of performing activities specified in this Master Declaration or any Supplemental Declaration.

**3.46. "Rate Stabilization Account"** means the Rate Stabilization Account of the Sewer Enterprise Fund established which is described in Section 4.5.

**3.47. "Rating Agency"** means Fitch, Moody's, S&P, or any other nationally recognized financial rating agency which has rated Outstanding Bonds or a Credit Facility at the request of the District.

**3.48. "Reciprocal Payment"** means scheduled payment to be made to, or for the benefit of the District under a Derivative Product by or on behalf of the Reciprocal Payor, which is either fixed in amount or is determined according to a formula set forth in the Derivative Product.

**3.49.** "**Reciprocal Payor**" means a party to a Derivative Product (other than the District) that is obligated to make one or more Reciprocal Payments thereunder, and which has at least an investment grade rating from a Rating Agency for its obligations under the Derivative Product.

**3.50. "Record Date"** for the Series 2002B Bonds, the Series 2009A Obligations, the Series 2009B Obligations, the Series 2010 Obligations, and the Series 2016 Bonds means the fifteenth (15th) day of the month preceding the month in which each Interest Payment Date occurs, whether or not a Business Day.

**3.51. "Registrar" or "Paying Agent"** means the paying agent and registrar designated by the District.

**3.52. "Reserve Credit Facility"** means any arrangement in which the District pays a fee in exchange for an agreement of a Credit Provider to advance money to the District in the future that the District will use to satisfy a reserve requirement for a subaccount in the Revenue Bond Reserve Account. "Reserve Credit Facility" does not include guaranteed investment contracts, master repurchase agreements and similar Permitted Investments.

**3.53. "Reserve Credit Facility Rating"** means a long-term debt, financial strength or claims paying ability rating assigned by a Rating Agency to a Reserve Credit Provider.

**3.54.** "Reserve Credit Provider" means a person or entity that is obligated to make payments under a Reserve Credit Facility, and any person or entity that has assumed, guaranteed, reinsured or otherwise become obligated to perform a Reserve Credit Provider's obligations to the District under a Reserve Credit Facility.

**3.55. "Revenue Bond Account"** means the Revenue Bond Account described in Section 5.2 of this Master Declaration.

**3.56. "Revenue Bond Reserve Account"** means the Revenue Bond Reserve Account in the Sewer Enterprise Fund described in Section 5.3 of this Master Declaration.

**3.57.** "S&P" means S&P Global Ratings, a corporation organized and existing under the laws of the State of New York, its successors and their assigns.

**3.58. "Separate Utility System"** means any utility property which is declared by the Board to constitute a system which is distinct from the Sewer System in accordance with Section 9.

**3.59.** "Series" refers to all Bonds or Parity Obligations authorized by a single Order and delivered in exchange for payment on the same date, regardless of variations in maturity, interest rate or other provisions, unless the closing documents for the Series provide otherwise.

**3.60.** "Series 2002B Bonds" means the District's Sewer System Refunding Revenue Bonds, Series 2002B, issued pursuant to this Master Declaration.

**3.61. "Series 2009A Obligations"** means the District's Sewer System Revenue Obligations, Series 2009A, issued pursuant to this Master Declaration.

**3.62.** "Series 2009B Obligations" means the District's Sewer System Revenue Obligations, Series 2009B, issued pursuant to this Master Declaration.

**3.63. "Series 2010 Obligations"** means the District's Sewer System Revenue Obligations, Series 2010, issued pursuant to this Master Declaration.

**3.64.** "Series 2016 Bonds" means the District's [Sewer System Revenue Bonds, Series 2016], issued pursuant to this Master Declaration.

**3.65. "Sewer Enterprise Fund"** means the collection of funds and accounts used by the District to account for the Gross Revenues and the proceeds of Bonds.

**3.66.** "Sewer System" means all real and personal property now or hereafter owned, operated, used, or maintained by the District for sanitary sewage disposal or sanitary sewage purification within or without the corporate limits of the District, including but not limited to, intercepting sewers, diversion sewers, relieving or interconnection sewers, lift stations, and plants for treatment, processing, and disposal of sanitary sewage. In addition, "Sewer System" includes all real and personal property now or hereafter owned, operated, used, or maintained by the District to provide surface water management, stormwater drainage or similar services to the North Clackamas Surface Water Management Area, as it now exists or may hereafter be expanded. However, the Sewer System does not include any Separate Utility System.

**3.67. "Stabilized Net Revenues"** means the Net Revenues for a period less deposits to the Rate Stabilization Account for the period, and plus withdrawals from the Rate Stabilization Account for the period.

**3.68.** "State" means the State of Oregon.

**3.69. "Subordinate Obligations"** means obligations having a lien on the Net Revenues that is subordinate and inferior to the lien of the Bonds. Restrictions on Subordinate Obligations are described in Section 8.

**3.70. "Subordinate Obligations Account"** means the Subordinate Obligations Account of the Sewer Enterprise Fund which is described in Section 5.5.

**3.71. "Supplemental Declaration"** means any declaration which supplements or amends this Master Declaration, entered into by the District in compliance with Section 13.

**3.72.** "Tax Maximum" means for any Series of Bonds, the lesser of the greatest amount of principal, interest and premium, if any, scheduled to be paid in any Fiscal Year on such Series; 125% of average amount of principal, interest and premium, if any, required to be paid on such Series during all Fiscal Years in which such Series will be Outstanding, calculated as of the date of issuance of such *Series;* or, ten percent of the proceeds of such Series, as "proceeds" is defined for purposes of Section 148(d) of the Code.

**3.73. "Valuation Date"** means July 1 of each year (or the first Business Day thereafter, if July 1 is not a Business Day), and the Business Day following any transfer from the Revenue Bond Reserve Account to the Revenue Bond Account pursuant to Section 5.4.1 or any similar subsection related to a subsequent subaccount in the Revenue Bond Reserve Account.

**3.74. "Variable Rate Obligations"** means any Bonds issued with a variable adjustable, convertible, or other similar interest rate which changes during the term of the Bonds, and any District Payments or Reciprocal Payments under a Parity Derivative Product for which the interest portion of the payment is based on a rate that changes during the term of the Derivative Product.

#### Section 4. Deposit, Pledge and Use of Gross Revenues

**4.1. Deposit of Gross Revenues.** All Gross Revenues shall be deposited when received in accounts and invested in securities which are not subject to the prior lien or claim of any person. All Gross Revenues shall be credited to the Sewer Enterprise Fund, and shall be used only as described in this Section as long as any Bonds remain Outstanding. Gross Revenues in the Sewer Enterprise Fund shall he used on or before the following dates for the following purposes in the following order of priority:

4.1.1. At any time to pay Operating Expenses which are then due.

4.1.2. Three Business Days prior to each Payment Date, the District shall credit Net Revenues to the Revenue Bond Account in an amount sufficient (with amounts available in the Revenue Bond Account) to pay in full all Bond principal, interest and premium, if any, which is due to be paid on that Payment Date.

4.1.3. On the first day of each month following a Valuation Date on which the balance in a subaccount of the Revenue Bond Reserve Account is determined to be less than its Reserve Requirement, the District shall credit to that subaccount the amount that is required to replenish that subaccount pursuant to Section 5.3.1.4. Section 5.4.7 sets the replenishment requirements for the First Reserve Subaccount.

4.1.4. On the day on which any rebates or penalties for Bonds are due to be paid to the United States pursuant to Section 148 of the Code, the District shall pay the amounts due from the Net Revenues.

4.1.5. On the dates specified in any proceedings authorizing Subordinate Obligations the District shall credit to the Subordinate Obligations Account the Net Revenues required by those proceedings.

4.1.6. On any date, the District may credit Net Revenues to the Rate Stabilization Account, apply Net Revenues to any franchise fees and similar charges imposed by the District or Clackamas County on the Sewer System or its operations, or spend Net Revenues for any other lawful purpose, but only if all credits and payments having a higher priority under this Section have been made.

**4.2. Pledge of Net Revenues.** The District hereby pledges the Net Revenues and any federal interest subsidies the District receives for Interest Subsidy Bonds to payment of principal of, premium (if any) and interest on all Bonds, and as security for such Bonds. In addition, the District hereby pledges the Net Revenues available for transfer to the Revenue Bond Reserve Account to pay amounts due under any Reserve Credit Facility. Pursuant to ORS 287A.310, these pledges of the Net Revenues hereby made by the District shall be valid and binding from the time of the adoption of this Master Declaration. The Net Revenues so pledged and hereafter received by the District shall immediately be subject to the lien of such pledge without any physical delivery or further act, and the lien of the pledge shall be superior to all other claims and liens whatsoever.

#### Section 5. Bond Funds and Accounts

**5.1. Required Accounts.** So long as Bonds are Outstanding, the District shall maintain the Revenue Bond Account, the Revenue Bond Reserve Account, and the Subordinate Obligations Account as discrete accounts in the Sewer Enterprise Fund.

**5.2. Revenue Bond Account.** The Revenue Bond Account shall be held and maintained by the District, and amounts credited to the Revenue Bond Account shall be deposited in accounts and invested in securities which are not subject to the prior lien or claim of any person. Until all Bonds are paid or defeased; amounts in the Revenue Bond Account shall be used only to pay Bonds. The District shall transfer sufficient amounts from the Revenue Bond Account to the Registrar in time to permit the Registrar to pay all Bond principal, interest and, premium (if any) when due in accordance with the Bonds. Amounts in the Revenue Bond Account shall be credited to the Revenue Bond Account.

#### 5.3. Revenue Bond Reserve Account.

5.3.1. The Revenue Bond Reserve Account shall be held by the District and the District may create subaccounts in the Revenue Bond Reserve Account to secure Bonds. When each subaccount is created, the District shall determine whether the subaccount will secure one or more Series of Bonds. If the District creates a subaccount in the Revenue Bond Reserve Account, the District shall, when it issues the first Series of Bonds that is secured by that subaccount:

5.3.1.1. establish the Reserve Requirement for that subaccount and pledge amounts credited to that subaccount to pay the Bonds that are secured by that subaccount;

5.3.1.2. determine if the Reserve Requirement for that subaccount may be funded with a Reserve Credit Facility and the requirements for such Reserve Credit Facility;

5.3.1.3. Establish the valuation requirements for that subaccount; and,

5.3.1.4. Establish the replenishment requirements for that subaccount.

5.3.2. The District shall not create any subaccounts in the Revenue Bond Reserve Account for any purpose except securing Bonds in accordance with this Master Declaration.

# **5.4.** The First Reserve Subaccount and the First Reserve Subaccount Reserve Requirement.

5.4.1. The First Reserve Subaccount is hereby created in the Revenue Bond Reserve Account. The First Reserve Subaccount shall secure only the Series 2002B Bonds, Series 2009A Obligations, Series 2009B Obligations, Series 2010 Obligations, Series 2016 Bonds and any subsequent Series of Bonds to which the amounts in the First Reserve Subaccount are pledged. Except as specifically provided in this Section 5.4, amounts credited to the First Reserve Subaccount shall be used only to pay principal, interest and premium, if any, on any Series of Bonds that are secured by the First Reserve Subaccount, and only if amounts in the Revenue Bond Account are not sufficient to make those payments.

5.4.2. The District hereby irrevocably pledges the amounts that are credited to the First Reserve Subaccount to pay the Series 2002B Bonds, Series 2009A Obligations, Series 2009B Obligations, Series 2010 Obligations, Series 2016 Bonds. Pursuant to ORS 287A.310, this pledge shall be valid and binding from the Closing date of the Series 2016 Bonds. The amounts so pledged and hereafter received by the District shall immediately be subject to the lien of this pledge without any physical delivery or further act, and the lien of this pledge shall be superior to all other claims and liens whatsoever to the fullest extent permitted by ORS 287A.310.

5.4.3. At Closing of the Series 2016 Bonds the District shall deposit into the First Reserve Subaccount an amount equal to the First Reserve Subaccount Reserve Requirement. The deposit may be made from Net Revenues, from Series 2016 Bond proceeds, from other amounts available to the District, or by crediting a Reserve Credit Facility to the First Reserve Subaccount. If the District elects to secure a Series of Parity Obligations with the First Reserve Subaccount, when the District issues that Series the District shall deposit an amount in the First Reserve Subaccount that is sufficient to make the balance in the First Reserve Subaccount at least equal to the First Reserve Subaccount Reserve Requirement, calculated with the Series of Parity Obligations treated as Outstanding.

5.4.4. At Closing of the 2016 Bonds, the First Reserve Subaccount Reserve Requirement is equal to \$\_\_\_\_\_\_ and is funded by \_\_\_\_\_\_.

5.4.5. The District covenants to maintain a balance in the First Reserve Subaccount which is equal to the First Reserve Subaccount Reserve Requirement, but solely from deposits of Net Revenues pursuant to Section 4.1.3, from the Closing deposit pursuant to Section 18.5.1.1., and similar Closing deposits for any subsequent Series of Bonds secured by the First Reserve Subaccount. However, the District reserves the right to make deposits to the First Reserve Subaccount from other sources.

5.4.6. The District shall value the First Reserve Subaccount on each First Reserve Subaccount Valuation Date. The value of the First Reserve Subaccount shall be equal to the

sum of the values of the Permitted Investments credited to the First Reserve Subaccount, calculated as provided in Section 5.4.12, plus the value of any Reserve Credit Facilities for the First Reserve Subaccount, calculated as provided in Section 5.4.10.

5.4.7. If the balance in the First Reserve Subaccount on a First Reserve Subaccount Valuation Date is less than the First Reserve Subaccount Reserve Requirement, the District shall begin making transfers of Net Revenues to the First Reserve Subaccount in accordance with Section 4.1.3.

5.4.7.1. If the deficiency results from a transfer from the First Reserve Subaccount or from the loss of value of Permitted Investments credited to the First Reserve Subaccount, each transfer required under Section 4.1.3 for the First Reserve Subaccount shall: 1) be equal to at least 1/12 of the deficiency discovered on the first First Reserve Subaccount Valuation Date on which the balance in the First Reserve Subaccount is less than the First Reserve Subaccount shall and 2) continue until the balance in the First Reserve Subaccount is equal to the First Reserve Subaccount Reserve Subaccount Reserve Subaccount Reserve Subaccount is equal to the First Reserve Subaccount Subaccount Subaccount Reserve Subaccou

5.4.7.2. If the deficiency results from a loss or withdrawal a Reserve Credit Facility Rating, each transfer required under Section 4.1.3 for the First Reserve Subaccount shall: 1) be equal to at least 1/60 of the deficiency discovered on the first First Reserve Subaccount Valuation Date on which the balance in the First Reserve Subaccount is less than the First Reserve Subaccount Requirement and 2) continue until the balance in the First Reserve Subaccount is equal to the First Reserve Subaccount Reserve Requirement.

5.4.8. If the balance in the First Reserve Subaccount on a First Reserve Subaccount Valuation Date is greater than the First Reserve Subaccount Reserve Requirement the District may transfer the excess to the Revenue Bond Account or the Subordinate Obligations Account. In addition the District may use the excess for any other purpose if the District obtains an opinion of Bond Counsel stating that the use of the excess for that purpose will not cause interest on any Bonds to become includable in gross income under the Code.

5.4.9. Earnings on the First Reserve Subaccount shall be credited to that subaccount whenever the balance in that subaccount is less than the First Reserve Subaccount Reserve Requirement. Otherwise, earnings shall be credited to the Revenue Bond Account.

5.4.10. Reserve Credit Facilities credited to the First Reserve Subaccount shall be valued on each First Reserve Subaccount Valuation Date in the following manner:

5.4.10.1. A Reserve Credit Facility shall be valued at the amount available to be drawn on it when a Rating Agency has a Reserve Credit Facility Rating in effect for the Reserve Credit Provider that is in one of the two highest rating categories, determined without regard to pluses, minuses or other degrees. Under rating systems in effect on the date of this Master Declaration, a long-term debt rating in one of the two highest rating categories by a Rating Agency would be a rating in the "AA" or "Aa" category or better. 5.4.10.2. A Reserve Credit Facility shall have no value when Rating Agency does not have a Reserve Credit Facility Rating in effect for the Reserve Credit Provider that is in one of the two highest rating categories, determined without regard to pluses, minuses or other degrees.

5.4.11. Moneys in the First Reserve Subaccount may be invested only in Permitted Investments that mature no later than the final maturity date of the Bonds that are secured by the First Reserve Subaccount, and to acquire Reserve Credit Facilities.

5.4.12. Permitted Investments in the First Reserve Subaccount shall be valued on each First Reserve Subaccount Valuation Date in the following manner:

5.4.12.1. Demand deposits, deposits in the Oregon Short Term Fund and other investments which mature in two years or less after the First Reserve Subaccount Valuation Date shall be valued at their face amount, plus accrued interest;

5.4.12.2. Investments which mature more than two years after the First Reserve Subaccount Valuation Date and for which bid and asked prices are published on a regular basis in the Wall Street Journal (or, if not there, then in the New York Times) shall be valued at the average of their most recently published bid and asked prices;

5.4.12.3. Investments which mature more than two years after the First Reserve Subaccount Valuation Date and for which the bid and asked prices are not published on a regular basis in the Wall Street Journal or the New York Times shall be valued at the average bid price quoted by any two nationally recognized government securities dealers (selected by the District in its absolute discretion) at the time making a market in such investments or the bid price published by a nationally recognized pricing service;

5.4.12.4. Certificates of deposit and bankers acceptances which mature more than two years after the First Reserve Subaccount Valuation Date shall be valued at their face amount, plus accrued interest; and

5.4.12.5. Any investment which is not specified above and which matures more than two years after the First Reserve Subaccount Valuation Date shall be valued at its fair market value as reasonably estimated by the District.

5.4.12.6. Any investment which is not specified above and which matures more than two years after the First Reserve Subaccount Valuation Date shall be valued at its fair market value as reasonably estimated by the District.

5.4.13. Withdrawals from the First Reserve Subaccount shall be made in the following order of priority:

5.4.13.1. First, from any cash on deposit in the First Reserve Subaccount; and

5.4.13.2. *Second*, from the liquidation proceeds of any Permitted Investments on deposit in such First Reserve Subaccount; and,

5.4.13.3. *Third*, from amounts drawn on Reserve Credit Facilities in the First Reserve Subaccount.

5.4.14. All amounts on deposit in the First Reserve Subaccount (other than amounts attributable to Reserve Credit Facilities) may be applied to the final payment (whether at maturity or by prior redemption) of the last remaining Series of Bonds secured by the First Reserve Subaccount. Amounts so applied shall be credited against the amounts the District is required to transfer into the Revenue Bond Account under Section 4.1.2.

5.4.15. Amounts in the First Reserve Subaccount (other than amounts attributable to Reserve Credit Facilities) may be transferred into escrow to defease Bonds secured by the First Reserve Subaccount, but only if the balance remaining in the First Reserve Subaccount after the transfer is at least equal to the First Reserve Subaccount Reserve Requirement calculated based on the Bonds that are secured by the First Reserve Subaccount which remain Outstanding after the defeasance.

**5.5. Subordinate Obligations Account.** The District has previously executed State Revolving Fund Loan Agreements in an approximate amount of \_\_\_\_\_\_, which constitute Subordinate Obligations. So long as any Subordinate Obligations are outstanding, the District shall maintain the Subordinate Obligations Account. The Subordinate Obligations Account may be divided into subaccounts, and the District may establish priorities for funding the subaccounts in the Subordinate Obligations Account. Net Revenues shall be credited to the Subordinate Obligations Account only as permitted by Section 4.1.5. Earnings on the Subordinate Obligations Account shall be credited as provided in the proceedings authorizing the Subordinate Obligations.

**5.6. Rate Stabilization Account.** The Rate Stabilization Account is hereby created within the Sewer Enterprise Fund The Rate Stabilization Account shall be held and maintained by the District, and amounts credited to the Rate Stabilization Account shall be deposited in accounts and invested in securities which are not subject to the prior lien or claim of any person. Net Revenues may be credited to the Rate Stabilization Account at the option of the District as permitted by Section 4.1.6. Money credited to the Rate Stabilization Account may be used. Amounts withdrawn from the Rate Stabilization Account increase Stabilized Net Revenues for the Fiscal Year in which they are withdrawn, and amounts credited to the Rate Stabilization Account reduce Stabilized Net Revenues for the Fiscal year in which they are credited. Credits to and withdrawals from the Rate Stabilization Account that occur within ninety days after the end of a Fiscal Year may be treated as occurring in the most recently ended Fiscal Year. Earnings on the Rate Stabilization Account shall be credited to the Sewer Enterprise Fund.

#### Section 6. Rate Covenant

**6.1. General Covenant.** The District covenants for the benefit of the Owners that it will establish and maintain rates and charges in connection with the ownership and operation of the Sewer System which are sufficient to permit the District to pay all Operating Expenses and all lawful charges against the Net Revenues, to remain in compliance with its duties under this Master Declaration and any Supplemental Declaration, and to make all transfers required by this

Master Declaration to the Revenue Bond Account, the Revenue Bond Reserve Account, and the Subordinate Obligations Account.

**6.2. Net Revenue Covenant.** The District covenants for the benefit of the Owners of all Bonds that it shall impose fees, rates and charges in connection with the ownership and operation of the Sewer System which, when combined with other Gross Revenues, are adequate:

6.2.1. To produce Net Revenues in each fiscal Year at least equal to one hundred ten percent (110.00%) of Annual Debt Service due in that Fiscal Year; and

6.2.2. To produce Stabilized Net Revenues each Fiscal Year at least equal to one hundred twenty percent (120.00%) of Annual Debt Service due in that Fiscal Year.

**6.3.** Compliance Report. Not later than ninety days after the end of each Fiscal Year the District shall file a certified report with the authorized Officer which demonstrates whether the District has complied with Section 6.2.1 and Section 6.2.2 during that Fiscal Year. If the report demonstrates that the District has not complied with Section 6.2.1 or Section 6.2.2 during that Fiscal Year, it shall not constitute an Event of Default if:

6.3.1. within one hundred twenty days after the end of the Fiscal Year, the District engages the services of a Qualified Consultant; and,

6.3.2. within one hundred eighty days after the end of the Fiscal Year, the Qualified Consultant recommends a schedule of fees, rates and charges or other actions which the Qualified Consultant reasonably projects will permit the District to comply with Section 6.2.1 and Section 6.2.2 for the remainder of the current Fiscal Year the next Fiscal Year; and,

6.3.3. within two hundred seventy days after the end of the Fiscal Year the District implements the recommendations of the Qualified Consultant.

**6.4. Interest Subsidy Bonds.** The amounts assumed to be paid on Interest Subsidy Bonds shall be calculated as follows:

6.4.1. When calculating Annual Debt Service for the rate covenant in Section 6.2, the District shall subtract from interest to be paid on Interest Subsidy Bonds the federal interest subsidies on Interest Subsidy Bonds that the District reasonably expects, at the beginning of the Fiscal Year, to receive during that Fiscal Year.

6.4.2. When calculating Annual Debt Service and Maximum Annual Debt Service for the tests for issuing Parity Obligations in Section 7, the District shall subtract from the scheduled payments of interest on Interest Subsidy Bonds the amount of federal interest subsidies that the District reasonably expects, at the time the Parity Obligations are issued, to receive.

6.4.3. When calculating the greatest amount of principal, interest and premium, if any, required to be paid in any Fiscal Year on a Series of Interest Subsidy Bonds to determine the Tax Maximum for Interest Subsidy Bonds that are secured by the First Reserve Subaccount, the District shall subtract from the scheduled payments of interest on Interest Subsidy Bonds

the federal interest subsidies that the District reasonably expects, at the time the Series of Interest Subsidy Bonds is issued, to be paid to the District for the Series of Interest Subsidy Bonds. The District shall not be required to increase the amount the District is required to hold in a subaccount in the Revenue Bond Reserve Account if federal interest subsidies are not paid when or in the amounts expected. However, if the District reduces the amount it holds in a subaccount of the Revenue Bond Reserve Account because Bonds secured by that subaccount have been paid, the District must take into account its reasonable expectations at the time of reduction in determining the amount that the District must retain in the First Reserve Subaccount.

#### Section 7. Parity Obligations

**7.1. Conditions for Issuance.** The District may issue Parity Obligations to provide funds for any purpose relating to the Sewer System, but only if:

7.1.1. No Event of Default under this Master Declaration or any Supplemental Declaration has occurred and is continuing.

7.1.2. At the time of the issuance of the Parity Obligations there is no deficiency in the Revenue Bond Account and the District has made all credits to the replenish subaccounts in the Revenue Bond Reserve Account that are required to have been by that time made pursuant to Section 4.1.3.

7.1.3. The covenant to impose fees, rates and charges in Section 6.2 of this Master Declaration shall apply to the proposed Parity Obligations, and any Supplemental Declaration authorizing the issuance those Parity Obligations shall contain a recital of that covenant.

7.1.4. There shall have been filed with the District either:

7.1.4.1. a certificate of an Authorized Officer stating that Net Revenues (adjusted as provided in the second sentence of this Section 7.2) for the Base Period were not less than one hundred twenty percent (120.00%) of the Maximum Annual Debt Service on all Outstanding Bonds, with the proposed Parity Obligations treated as Outstanding; or,

7.1.4.2. a certificate or opinion of a Qualified Consultant stating:

7.1.4.2.1. the amount of the Adjusted Net Revenues computed as provided in Section 7.3 below; and,

7.1.4.2.2. that the amount shown in Section 7.1.4.2.1 is not less than the sum of one hundred twenty percent (120.00%) of the Maximum Annual Debt Service on all Outstanding Bonds, with the Proposed Parity Obligations treated as Outstanding.

**7.2.** Adjustment of Historical Revenues. Net Revenues may be adjusted for purposes of Section 7.1.4.1 by adding any Net Revenues the Authorized Officer calculates the District would have had during the Base Period because of increases in Sewer System rates, fees and charges which took effect after the beginning of the Base Period. However, no adjustment shall be made

for these increases unless they have been approved by the Board prior to delivery of the Proposed Parity Obligations and are required to take effect no later than sixty days after the delivery of the proposed Parity Obligations.

**7.3.** Adjustment of Projected Revenues. Adjusted Net Revenues for purposes of Section 7.1.4.2 shall be computed by adjusting the Net Revenues for the Base Period as provided in this Section 7.3:

7.3.1. The District shall provide the Qualified Consultant with the following information:

7.3.1.1. The Base Period and the Net Revenues for the Base Period;

7.3.1.2. Information regarding any Sewer System utility properties that are being acquired with Parity Obligations and have an earnings record;

7.3.1.3. Any changes in rates and charges which have been adopted by the District since the beginning of the Base Period and the dates on which they are scheduled to take effect;

7.3.1.4. Any changes in customers since the beginning of the Base Period; and,

7.3.1.5. A description of any extensions or additions to the Sewer System that were in the process of construction at the beginning of the Base Period or commenced construction after the beginning of the Base Period, the expected date of completion of those extensions or additions, the estimated operating and capital costs of those extensions or additions, and any other changes to the Gross Revenues or Operating Expenses that the District reasonably expects to result from the completion and operation of those extensions or additions.

7.3.2. Using the information provided by the District pursuant to Section 7.3.1 and any additional information the Qualified Consultant determines is necessary, the Qualified Consultant may adjust the Net Revenues for the Base Period:

7.3.2.1. To reflect any changes that the Qualified Consultant projects will result from the acquisition of Sewer System utility properties that are being financed with the Parity Bonds and that have an earnings record;

7.3.2.2. To reflect any changes in rates and charges which have been adopted by the District and which have taken effect or are scheduled to take effect during the twelve months after the date of the Qualified Consultant's certificate, or which increase rates and charges for inflation at a level which the Qualified Consultant determines is reasonable;

7.3.2.3. To reflect any changes in customers of the Water System that occurred after the beginning of the Base Period and prior to the date of the Qualified Consultant's certificate; and

7.3.2.4. To reflect any changes to Gross Revenues or Operating Expenses not included in the preceding paragraphs that are projected to result from the completion and operation of additions and extensions to the Sewer System that were under construction at the

beginning of the Base Period, or commenced construction after the beginning of the Base Period.

**7.4. Refunding Exception.** The District may issue Parity Obligations to refund Outstanding Bonds without complying with Sections 7.1.1, 7.1.2, 7.1.3 and 7.1.4 if the refunded Bonds are defeased on the date of delivery of the refunding Parity Obligations and if the Annual Debt Service on the refunding Parity Obligations does not exceed the Annual Debt Service on the refunded Bonds in any Fiscal Year by more than \$5,000.

**7.5. Derivative Products as Parity Obligations.** A Derivative Product may be a Parity Derivative Product and a Parity Obligation if the obligation to make District Payments under the Derivative Product qualifies as a Parity Obligation under Section 7.1, after the Reciprocal Payments under the Derivative Product are applied to reduce Annual Debt Service. Any Parity Derivative Product shall clearly state that it is a Parity Derivative Product and has qualified as a Parity Obligation under Section 7 of this Master Declaration. The District shall credit any Reciprocal Payments from a Parity Derivative Product directly to the Revenue Bond Account.

**7.6. Lien of Parity Obligations.** Each Series of Parity Obligations issued in accordance with this Master Declaration shall have a lien on the Net Revenues which is equal to the lien of all other Outstanding Bonds.

#### Section 8. Subordinate Obligations

The District has previously entered into State Revolving Fund loan agreements; any portion of those loan agreements that remain outstanding after the 2016 Bonds are issued will be subordinated to the Bonds and shall constitute Subordinate Obligations. The District may issue Subordinate Obligations in the future only if:

**8.1. Payment Limited.** The Subordinate Obligations are payable solely from amounts permitted to be credited to the Subordinate Obligations Account pursuant to Section 4.1.6; and,

**8.2. Statement of Limitation.** The Subordinate Obligations state clearly that they are secured by a lien on or pledge of the Net Revenues which is subordinate and inferior to the lien on, and pledge of, the Net Revenues for the Bonds.

#### Section 9. Separate Utility System

The District may declare property which the District owns and is part of the Sewer System (but has a value of less than five percent of the Sewer System at the time of the declaration), and property which the District has not yet acquired but would otherwise become part of the Sewer System, to be part of a Separate Utility System. The District may pay costs of acquiring, operating and maintaining Separate Utility Systems from Net Revenues, but only if there is no deficit in the Revenue Bond Account or the Revenue Bond Reserve Account. The District may issue obligations which are secured by the revenues to pay the Separate Utility System, and may pledge the Separate Utility System revenues to pay those obligations. In addition, the District may issue Subordinate Obligations to pay for costs of a Separate Utility System, and may pledge the revenues of the Separate Utility System to pay the Subordinate Obligations.

#### Section 10. General Covenants

The District hereby covenants and agrees with the Owners of all Outstanding Bonds as follows:

**10.1. Payment of Bonds.** That it will promptly cause the principal, premium, if any, and interest on the Bonds to be paid as they become due in accordance with the provisions of this Master Declaration and any Supplemental Declaration.

**10.2.** Books and Records; Accounting for Revenues. That it will maintain complete books and records relating to the operation of the Sewer System and all District funds and accounts in accordance with generally accepted accounting principles applicable to municipal enterprises such as the District, and will cause such books and records to be audited annually at the end of each Fiscal Year, and an audit report prepared by the Auditor and made available for the inspection of Bondowners. Gross Revenues, Operating Expenses, Net Revenues and all similar terms shall be measured and accounted for in accordance with generally accepted accounting principles applicable to municipal enterprises such as the District, and not on a cash basis.

**10.3.** No Superior Obligations. That it will not issue Bonds or other obligations having a claim superior to the claim of the Bonds upon the Net Revenues.

**10.4. Prompt Deposit and Credit.** That it will promptly deposit and credit to all funds and accounts all sums required to be so deposited and credited.

**10.5. Operation of System.** That it will operate the Sewer System in a sound, efficient and economic manner, that it will not enter into any agreement to provide Sewer System products or services at a discount from published rate schedules, and that it will not provide free Sewer System products or services except for fire suppression and in case of emergencies.

**10.6. Insurance.** That it will at all times maintain with responsible insurers all such insurance on the Sewer System as is customarily maintained with respect to works and properties of like character against accident to, loss of or damage to such works or properties.

10.6.1. The net proceeds of insurance against accident to or destruction of the Sewer System shall be used first to repair or rebuild the damaged or destroyed Sewer System, and to the extent excess insurance proceeds remain, shall be applied to the payment or redemption of the Bonds on a pro rata basis.

10.6.2. Insurance described in Section 10.6 shall be in the form of policies or contracts for insurance with insurers of good standing and shall be payable to the District, or in the form of self-insurance by the District. The District shall establish such fund or funds or reserves which it deems are necessary to provide for its share of any such self-insurance.

**10.7.** Alienation of System Property. The District will not, nor will it permit others to, sell, mortgage, lease, transfer, assign, convey or otherwise dispose of or encumber all or any portion of the Sewer System except:

10.7.1. The District may dispose of all or substantially all of the Sewer System, only if the District pays all Bonds or defeases them pursuant to Section 14.

10.7.2. Except as provided in Section 10.7.3, the District will not dispose of any part of the Sewer System in excess of 5% of the aggregate market value of the Sewer System then in service unless prior to such disposition either:

10.7.2.1. there has been filed with the District a certificate of a Qualified Consultant stating that such disposition will not impair the ability of the District to comply with the rate covenants contained in Section 6.1 of this Master Declaration; or

10.7.2.2. provision is made for the payment, redemption or defeasance of a principal amount of Bonds equal to the greater of the following amounts:

10.7.2.2.1. An amount which will be in the same proportion to the net principal amount of Bonds then Outstanding (defined as the total principal amount of Bonds then Outstanding less the amount of cash and investments in the Revenue Bond Debt Service Fund and the Revenue Bond Reserve Account allocable to such Bonds) that the Gross Revenues attributable to the part of the Sewer System sold or disposed of for the 12 preceding months bears to the total Gross Revenues for such period; or

10.7.2.2.2. An amount which will be in the same proportion to the net principal amount of Bonds then Outstanding that the book value of the part of the Sewer System sold or disposed of bears to the aggregate book value of the Sewer System immediately prior to such sale or disposition.

10.7.3. The District may dispose of any portion of the Sewer System that has become unserviceable, inadequate, obsolete, or unfit to be used or no longer necessary for use in the operation of the Sewer System.

10.7.4. If the ownership of all or part of the Sewer System is transferred from the District through the operation of law, the District shall to the extent authorized by law, reconstruct or replace such transferred portion using any proceeds of the transfer unless the Board reasonably determines that such reconstruction or replacement is not in the best interest of the District and the Bondowners, in which case any proceeds shall be used for the payment, redemption or defeasance of the Bonds.

#### Section 11. Events of Default and Remedies

**11.1. Continuous Operation Essential.** The Board hereby finds and determines that the continuous operation of the Sewer System and the collection, deposit, credit and disbursement of the Net Revenues in the manner provided in this Master Declaration and in any Supplemental Declaration are essential to the payment and security of the Bonds, and the failure or refusal of the District to perform the covenants and obligations contained in this Master Declaration or any such Supplemental Declaration will endanger the necessary continuous operation of the Sewer System and the application of the Net Revenues to the operation of the Sewer System and the payment of the Bonds.

**11.2. Events of Default.** The following shall constitute "Events of Default":

11.2.1. If the District shall fail to pay any Bond principal or interest when due, either at maturity, upon exercise of a right of tender in the case of Variable Rate Obligations, by proceedings for mandatory or optional redemption or otherwise;

11.2.2. Except as provided in Sections 6.3 and 11.3, if the District shall default in the observance and performance of any other of its covenants, conditions and agreements in this Master Declaration, if such default continues for ninety (90) days after the District receives a written notice, specifying the Event of Default and demanding the cure of such default, from the Bondowners Committee or from the Owners of not less than twenty percent (20%) in aggregate principal amount of the Bonds Outstanding;

11.2.3. If the District shall sell, mortgage, lease, transfer, assign, convey or otherwise dispose of or encumber any properties constituting the Sewer System in violation of Section 10.7;

11.2.4. If an order, judgment or decree shall be entered by any court of competent jurisdiction:

11.2.4.1. appointing a receiver, trustee or liquidator for the District or the whole or any part of the Sewer System;

11.2.4.2. approving a petition seeking a declaration of bankruptcy, or the arrangement or reorganization of the District under any applicable law of the United States or the State; or

11.2.4.3. assuming custody or control of the District or of the whole or any part of the Sewer System under the provisions of any other law for the relief or aid of debtors and such order, judgment or decree shall not be vacated or set aside or stayed (or, in case custody or control is assumed by said order, such custody or control shall not be otherwise terminated) within sixty (60) days from the date of the entry of such order, judgment or decree; or

11.2.5. If the District shall:

11.2.5.1. admit in writing its inability to pay its debts generally as they become due;

11.2.5.2. file a petition in bankruptcy or seeking a composition of indebtedness under any state or federal bankruptcy or insolvency law;

11.2.5.3. make an assignment for the benefit of its creditors;

11.2.5.4. consent to the appointment of a receiver of the whole or any part of the Sewer System; or

11.2.5.5. consent to the assumption by any court of competent jurisdiction under the provisions of any other law for the relief or aid of debtors of custody or control of the District or of the whole or any part of the Sewer System.

**11.3. Exception.** It shall not constitute an Event of Default under Section 11.2.2 if the default cannot practicably be remedied within ninety days after the District receives notice of the default, so long as the District promptly commences reasonable action to remedy the default after the notice is received, and continues reasonable action to remedy the default until the default is fully and actually remedied.

**11.4. Remedies.** If an Event of Default occurs, any Bondowner may exercise any remedy available at law or in equity. However, the Bonds shall not be subject to acceleration.

**11.5. Appointment of Trustee.** During the continuance of an Event of Default described in Section 11.2, the owners of twenty percent (20%) in aggregate principal amount of the Bonds then outstanding may call a Bondowners meeting for the purpose of electing a Bondowners Committee.

11.5.1. Such meeting shall be called and the proceedings thereof shall be conducted in the manner provided in Section 12 hereof.

11.5.2. At such meeting the Bondowners present in person or by proxy may, by a majority of the votes cast, elect one or more persons, who may or may not be Bondowners, to the Bondowners Committee which shall act as trustee for all Bondowners, and the Bondowners Committee as such trustee may have and exercise all the rights and powers provided for in this Master Declaration to be exercised by the Bondowners Committee. The Bondowners present in person or by proxy at said meeting, or at any adjourned meeting thereof, shall prescribe the manner in which the successors of the persons elected to the Bondowners Committee at such Bondowners meeting shall be elected or appointed, and may prescribe rules and regulations governing the exercise by the Bondowners Committee of the powers conferred upon it herein, and may provide for the termination of the existence of the Bondowners in the manner herein provided, and their successors, as a committee are hereby declared to be trustees for the owners of all the Bonds then outstanding, and are empowered to exercise in the name of the Bondowners Committee.

#### 11.6. Books of District Open to Inspection.

11.6.1. The District covenants that if an Event of Default shall have happened and shall not have been remedied, the books of record and account of the District and all other records relating to the Sewer System shall at all reasonable times be subject to the inspection and use of the Bondowners Committee and any persons holding at least twenty percent (20%) of the principal amount of Outstanding Bonds and their respective agents and attorneys.

11.6.2. The District covenants that if the Event of Default shall happen and shall not have been remedied, the District will continue to account, as a trustee of an express trust, for all Net Revenues and other moneys, securities and funds pledged under the Master Declaration.

**11.7. Payment of Funds to Bondowners Committee.** The District covenants that if an Event of Default shall happen and shall not have been remedied, the District upon demand of the

Bondowners Committee, shall, if it is then lawful to do so, pay over to the Bondowners Committee:

11.7.1. forthwith, all moneys, securities and funds then held by the District and pledged under the Master Declaration, and

11.7.2. as promptly as practicable after receipt thereof, all Gross Revenues.

11.8. Possession by Bondowners Committee of Properties of Sewer System; Receivership. At any time after the occurrence of an Event of Default and prior to the curing of such Event of Default the Bondowners Committee, as a matter of right against the District, shall, to the extent permitted by law, be entitled to take possession and control of the business and properties of the Sewer System. Upon taking such possession, the Bondowners Committee shall operate and maintain the Sewer System, make any necessary repairs, renewals and replacements in respect thereof, prescribe rates and charges for the Sewer System, collect the Gross Revenues, and perform all of the agreements and covenants contained in all contracts which the District is at the time obligated to perform. At any such time and if permitted by law the Bondowners Committee shall be entitled to the appointment of a receiver of the business and property of the Sewer System, of the moneys, securities and funds of the District pledged under the Master Declaration, and of the Gross Revenues, and of the income therefrom, with all such powers as the court or courts making such appointment shall confer, including the power to perform and enforce all contracts, to the same extent that the District shall then be entitled and obligated to do. Notwithstanding the appointment of any receiver, the Bondowners Committee shall be entitled to retain possession and control of and to collect and receive income from any moneys, securities, funds and Gross Revenues deposited, credited or pledged with or to it under the Master Declaration or agreed or provided to be delivered to or deposited, credited or pledged with or to it under the Master Declaration.

## 11.9. Application of Funds by Bondowners Committee.

11.9.1. During the continuance of an Event of Default, the Gross Revenues received by the Bondowners Committee, whether pursuant to the provisions of the preceding paragraph, or as the result of taking possession of the business and properties of the Sewer System, shall be applied by the Bondowners Committee, first, to the payment of the reasonable and proper charges, expenses and liabilities paid or incurred by the Bondowners Committee (including the cost of securing the services of any engineer or firm of engineers selected for the purpose of rendering advice with respect to the operation, maintenance, repair and replacement of the Sewer System necessary to prevent any loss of Gross Revenues, and with respect to the sufficiency of the rates and charges for services and products sold, furnished or supplied by the Sewer System), second, to the payment of the Operating Expenses, and third to the payment of the principal of, premium, if any, and interest on the Bonds.

11.9.2. In the event that at any time the funds held by the Bondowners Committee and the Paying Agents for the Bonds shall be insufficient for the payment of the principal of, premium, if any, and interest then due on the Bonds, such funds (other than funds held for the payment or redemption of particular Bonds which have theretofore become due at maturity or by call for redemption) and all Gross Revenues and other moneys received or collected for

the benefit or for the account of owners of the Bonds by the Bondowners Committee shall be applied as follows:

11.9.2.1. **First,** to the payment to the persons entitled thereto of all installments of interest then due on Bonds in the order of the maturity of such installments, earliest maturities first, and, if the amount available shall not be sufficient to pay in full any installment or installments or interest maturing on the same date, then to the payment thereof ratably, according to the amounts due thereon, to the persons entitled thereto, without any discrimination or preference; and

11.9.2.2. **Second,** to the payment to the persons entitled thereto of the unpaid principal and premium, if any, of any Bonds which shall have become due, whether at maturity or by call for redemption, in the order of their due dates, earliest maturities first, and, if the amount available shall not be sufficient to pay in full all the Bonds due on any date, then to the payment thereof ratably, according to the amounts of principal and premium, if any, due on such date, to the persons entitled thereto, without any discrimination or preference.

11.10. Relinquishment of Possession and Funds Upon Remedy of Default. If and whenever all overdue installments of interest on all Bonds, together with the reasonable and proper charges, expenses, and liabilities of the Bondowners Committee and the owners of Bonds, their respective agents and attorneys, and all other sums payable by the District under the Master Declaration including the principal of, premium, if any, and accrued unpaid interest on all Bonds which shall then be payable, shall either be paid by or for the account of the District, or provision satisfactory to the Bondowners Committee shall be made for such payment, and all defaults under the Master Declaration or the Bonds shall be made good or secured to the satisfaction of the Bondowners Committee or provision deemed by the Bondowners Committee to be adequate shall be made therefor, the Bondowners Committee shall relinquish possession and control of the Sewer System and pay over to the District all moneys, securities, funds and Gross Revenues then remaining unexpended in the hands of the Bondowners Committee and thereupon all Gross Revenues shall thereafter be applied as provided in Section 9 of this Master Declaration. No such payment over to the District by the Bondowners Committee or resumption of the application of Gross Revenues as provided in Section 11 of this Master Declaration shall extend to or affect any subsequent default under the Master Declaration or impair any right consequent thereon.

#### 11.11. Suits at Law or in Equity.

11.11.1. If an Event of Default shall happen and shall not have been remedied, then and in every such case, the Bondowners Committee by its agents and attorneys, shall be entitled and empowered to proceed forthwith to take such necessary steps and institute such suits, actions and proceedings at law or in equity for the collection of all sums due in connection with the Bonds and to protect and enforce the rights of the owners of the Bonds under the Master Declaration, for the specific performance of any covenant herein contained or in aid of the execution of any power herein granted, or for an accounting against the District as trustee of an express trust, or in the enforcement of any other legal or equitable right as the

Bondowners Committee, being advised by counsel, shall deem most effectual to enforce any of the rights of the owners of the Bonds.

11.11.2. Any action, suit or other proceedings instituted by the Bondowners Committee hereunder shall be brought in its name as trustee for the Bondowners and all such rights of action upon or under any of the Bonds or the provisions of this Master Declaration may be enforced by the Bondowners Committee without the possession of any of said Bonds, and without the production of the same at any trial or proceedings relative thereto except where otherwise required by law, and the respective owners of said Bonds, by taking and holding the same, shall be conclusively deemed irrevocably to appoint the Bondowners Committee the true and lawful trustee of the respective owners of said Bonds, with authority to institute any such action, suit or proceeding; to receive as trustee and deposit in trust any sums becoming distributable on account of said Bonds; to execute any paper or documents for the receipt of such moneys, and to do all acts with respect thereto that the Bondowner himself might have done in person, provided however, that nothing herein contained shall be deemed to authorize or empower the Bondowners Committee to consent to, accept or adopt, on behalf of any owner of Bonds, any plan of reorganization or adjustment affecting the said Bonds of the District or any right of any owner thereof; or to authorize or empower the Bondowners Committee to vote the claims of the owners hereof in any receivership, insolvency, liquidation, bankruptcy, reorganization or other proceeding to which the District shall be a party; and provided further, however, that any Bondowner or Bondowners may by mutual agreement transfer title to the Bonds held by him or them to the Bondowners Committee, or may by agreement with other Bondowners create or organize a separate trustee or Bondowners Committee and may confer upon the Bondowners Committee or such separate trustee or Bondowners Committee, such powers and duties as such agreement or agreements shall provide, and the provisions of this Master Declaration shall not be construed as a limitation on the powers and duties which consenting Bondowners may by agreement confer on the Bondowners Committee or such separate trustee or Bondowners Committee. The Bondowners Committee shall have full power of substitution and delegation in respect to any of the powers hereby granted.

**11.12. Direction of Actions of Bondowners Committee.** The owners of not less than a majority in aggregate principal amount of the Bonds that are the subject of a Bondowners Committee at the time Outstanding, may direct the time, method and place of conducting any proceeding for any remedy available to the Bondowners Committee, or exercising any trust or power conferred upon the Bondowners Committee, provided that the Bondowners Committee shall be provided with reasonable security and indemnity and shall have the right to decline to follow any such direction only (i) if the Bondowners Committee shall be advised by counsel that the action or proceeding so directed may not lawfully be taken; or (ii) if the Bondowners Committee in good faith shall determine that the action or proceeding so directed would involve the Bondowners Committee in personal liability or that the action or proceeding so directed would be unjustly prejudicial to the owners of Bonds not parties to such direction.

**11.13. Suits by Individual Bondowners.** No owner of any one or more of the Bonds shall have any right to institute any action, suit or proceeding at law or in equity for the enforcement of any provision of the Master Declaration or the execution of any trust under the Master Declaration or for any remedy under the Master Declaration, unless an Event of Default shall have happened

and be continuing, and unless no Bondowners Committee has been created as herein provided, but unless no Bondowners Committee has been created as herein provided, but any remedy herein authorized to be exercised by the Bondowners Committee, except the right to take possession of the Gross Revenues and properties of the Sewer System, but including the right to the appointment of a receiver of the business and properties of the Sewer System, may be exercised individually by any Bondowner, in his own name and on his own behalf or for the benefit of all Bondowners, in the event no Bondowners Committee has been created, or with the consent of the Bondowners Committee, if such Bondowners Committee has been created; provided, however, that nothing contained in the Master Declaration or in the Bonds shall affect or impair the obligation of the District, which is absolute and unconditional, to pay at the respective dates of maturity and places therein expressed the principal of, premium, if any, and interest on the Bonds to the respective owners thereof, or affect or impair the rights of action, which are also absolute and unconditional, of any owner to enforce the payment of this Bonds, or to reduce to judgment his claim against the District for the payment of the principal of and interest on his Bonds, without reference to, or the consent of, the Bondowners Committee or any other owner of Bonds.

#### 11.14. Waivers of Default.

11.14.1. No delay or omission of the Bondowners Committee or of any owner of Bonds to exercise any right or power arising upon the happening of an Event of Default shall impair any right or power or shall be construed to be a waiver of any such Event of Default or to be an acquiescence therein; and every power and remedy given by this Article to the Bondowners Committee or to the owners of Bonds may be exercised from time to time and as often as may be deemed expedient by the Bondowners Committee or by such owners.

11.14.2. The Bondowners Committee or the owners of not less than fifty percent (50%) in principal amount of the Bonds that are the subject of the Bondowners Committee and are at the time Outstanding, or their attorneys-in-fact duly authorized, may on behalf of the owners of all of the Bonds that are the subject of the Bondowners Committee waive any past default under the Master Declaration with respect to such Bonds and its consequences, except a default in the payment of the principal of, premium, if any, or interest on any of the Bonds. No such waiver shall extend to any subsequent or other default or impair any right consequent thereon.

**11.15. Remedies Granted in Master Declaration Not Exclusive.** No remedy by the terms of the Master Declaration conferred upon or reserved to the Bondowners Committee or the owners of the Bonds is intended to be exclusive of any other remedy, but each and every such remedy shall be cumulative and shall be in addition to every other remedy given under the Master Declaration or existing at law or in equity or by statute on or after the date of adoption of the Master Declaration.

#### Section 12. Bondowners Meetings

**12.1. Call of Bondowners Meetings.** The District, the Bondowners Committee or the owners of not less than twenty percent (20%) in aggregate principal amount of the Bonds then Outstanding may at any time call a meeting of the owners of the Bonds. Every such meeting

shall be held at such place in the City of New York, State of New York, or in the City of Portland, State of Oregon, as may be specified in the notice calling such meeting. Written notice of such meeting, stating the place and time of the meeting and in general terms the business to be transacted, shall be mailed to the Bondowners by the District, the Bondowners Committee or the Bondowners calling such meeting not less than thirty (30) nor more than sixty (60) days before such meeting, and shall be published at least once a week for four (4) successive calendar weeks on any day of the week, the date of first publication to be not less than thirty (30) nor more than sixty (60) days preceding the meeting; provided, however, that the mailing of such notice shall in no case be a condition precedent to the validity of any action taken at any such meeting. The expenses of publication of such notice shall be paid or reimbursed by the District. The list of Bondholders' names & addresses maintained by the Registrar may only be released by the District. Any meeting of Bondowners shall, however, be valid without notice if the owners of all Bonds then Outstanding are present in person or by proxy or if notice is waived before or within thirty (30) days after the meeting by those not so present.

**12.2.** Notice to Bondowners. Except as otherwise provided in this Master Declaration, any provision in this Master Declaration for the mailing of a notice or other paper to Bondowners shall be fully complied with if it is mailed by first class mail, postage prepaid, to each registered owner of any of the Bonds then outstanding at his address, if any, appearing upon the Revenue Bond Register; and any provision in this Master Declaration contained for publication of a notice or other matter shall require the publication thereof in *"The Daily Bond Buyer"* in the City of New York, State of New York (or in lieu of publication in *"The Daily Bond Buyer,"* in a daily newspaper printed in the English language and customarily published on each business day of general circulation in the English language and customarily published on each business day and of general circulation in the City of Portland, State of Oregon.

## 12.3. Proxies; Proof of Ownership of Bonds.

12.3.1. Attendance and voting by Bondowners at such meetings may be in person or by proxy. Owners of Bonds may, by an instrument in writing under their hands, appoint any person or persons, with full power and substitution, as their proxy to vote at any meeting for them. Officers or nominees of the District may be present or represented at such meeting and take part therein but shall not be entitled to vote thereat, except as such officers or nominees are Bondowners or proxies for Bondowners.

12.3.2. Any registered owner of Bonds shall be entitled in person or by proxy to attend and vote at such meeting as owner of the Bonds registered in his name without producing such Bonds, and such persons and their proxies shall, if required, produce such proof of personal identity as shall be satisfactory to the Secretary of the meeting. All proxies presented at such meeting shall be delivered to the Inspectors of Votes and filed with the Secretary of the meeting.

12.3.3. The vote at any such meeting of the owner of any Bond entitled to vote thereat shall be binding upon such owner and upon every such subsequent owner of such Bond (whether or not such subsequent owner has notice thereof).

#### 12.4. Execution of Instruments by Bondowners.

12.4.1. Any request, direction, consent or other instrument in writing required or permitted by this Master Declaration to be signed or executed by Bondowners may be in any number of concurrent instruments of similar tenor, and may be signed or executed by such Bondowners in person or by agent appointed by an instrument in writing. Proof of the execution of any such instrument shall be sufficient for any purpose of this Master Declaration if made by either

12.4.1.1. an acknowledgment executed by a notary public or other officer empowered to take acknowledgments of deeds to be recorded in the particular jurisdiction, or

12.4.1.2. an affidavit of a witness to such execution sworn to before such a notary public or other officer.

12.4.2. Where such execution is by an officer of a corporation or association or a member of a partnership on behalf of such corporation, association or partnership, such acknowledgment or affidavit shall also constitute sufficient proof of his authority.

12.4.3. The foregoing shall not be construed as limiting the District to such proof, it being intended that the District may accept any other evidence of the matters herein stated which it may deem sufficient. Any request or consent of the owner of any Bond shall bind every future owner of the same Bond in respect of anything done by the District in pursuance of such request, direction or consent.

12.4.4. The right of a proxy for a Bondowner to act may be proved (subject to the District's right to require additional proof) by a written proxy executed by such Bondowner as aforesaid.

**12.5. Appointment of Officers at Bondowners Meetings.** Persons named by the District or elected by the owners of a majority in principal amount of the Outstanding Bonds represented at the meeting in person or by proxy in the event the District is not represented at such meeting, shall act as temporary Chairman and temporary Secretary of any meeting of Bondowners. A permanent Chairman and a permanent Secretary of such meeting shall be elected by the owners of a majority in principal amount of the Bonds represented at such meeting in person or by proxy. The permanent chairman of the meeting shall appoint two (2) Inspectors of Votes who shall count all votes cast at such meeting, except votes on the election of Chairman and Secretary as aforesaid, and who shall make and file with the Secretary of the meeting and with the District their verified report of all such votes cast at the meeting.

**12.6.** Quorum at Bondowners Meetings. The owners of not less than the principal amount of the Bonds required for any action to be taken at such meeting must be present at such meeting in person or by proxy to constitute a quorum for the transaction of business, less than a quorum, however, having power to adjourn from time to time without any other notice than the announcement thereof at the meeting; provided, however, that, if such meeting is adjourned by less than a quorum for more than ten (10) days, notice thereof shall be published by the District at least five (5) days prior to the adjourned dated of the meeting.

**12.7. Vote Required to Amend Master Declaration.** Any amendment to the provisions of the Master Declaration, in any particular except the percentage of Bondowners the approval of which is required to approve such amendment, may be made by a Supplemental Declaration of the District and an order duly adopted by the affirmative vote at a meeting of Bondowners duly convened and held, or with written consent as hereinafter provided in this Section 12.7, of the owners of not less than fifty-one percent (51%) in principal amount of the Bonds Outstanding when such meeting is held or such consent is given; provided, however, that no such amendment shall:

12.7.1. extend the date of payment of the principal of any Bond or of any installment of interest thereon or reduce the principal or redemption price thereof or the rate of interest thereon or advance the date upon which any Bond may first be called for redemption prior to its fixed maturity date;

12.7.2. give to any Bond or Bonds any preference over any other Bond or Bonds secured equally and ratably therewith;

12.7.3. reduce the aforesaid percentage of Bonds, the owners of which are required to consent to any such order amending the provisions of this Master Declaration; or

12.7.4. authorize the creation of any pledge superior to or, except as provided in Section 7 for the issuance of Parity Obligations, on a parity with the pledge of Net Revenues afforded by this Master Declaration for Bonds, without the consent of the owner of each such Bond affected thereby.

12.8. Obtaining Approval of Amendments at Bondowners Meeting. The District may at any time adopt an order amending the provisions of the Master Declaration to the extent that such amendment is permitted by the provisions of Section 13 hereof, to take effect when and as provided in this Section 12. At any time thereafter such order may be submitted by the District for approval to a meeting of the Bondowners duly convened and held in accordance with the provisions of the Master Declaration. A record in duplicate of the proceedings of each meeting of the Bondowners shall be prepared by the permanent secretary of the meeting and shall have attached thereto the original reports of the inspectors of votes and affidavits by a person or persons having knowledge of the facts, showing a copy of the notice of the meeting and setting forth the facts with respect to the mailing and publication thereof under the provisions of the Master Declaration. Such a record shall be signed and verified by the affidavits of the permanent chairman and the permanent secretary of the meeting, and one duplicate thereof shall be delivered to the District. Any record so signed and verified shall be proof of the matters therein stated. If the order of the District making such amendment shall be approved by an order duly adopted at such meeting of Bondowners by the affirmative vote of the owners of the required percentages of Bonds, a notice stating that an order approving such amendment has been so adopted shall be mailed by the District to each Bondowner who has requested such notice (but failure so to mail copies of such notice shall not affect the validity of such order) and shall be published at least once in the manner provided for publication of notices of redemption of Bonds. Proof of such mailing and publication by the affidavit or affidavits of a person or persons having knowledge of the facts shall be filed with the District. Such order of the District making such amendment shall be deemed conclusively to be binding upon the District, the Paying Agent, and

the Owners of all Bonds at the expiration of thirty (30) days after the publication of the notice provided for in this Section 12, except in the event of a final decree of court of competent jurisdiction setting aside such order or annulling the action taken thereby in a legal action or equitable proceeding for such purpose commenced within such period; provided that the District and any Paying Agent during such thirty (30) day period and any such further period during which such action or proceeding may be pending shall be entitled in their absolute discretion to take such action, or to refrain from taking such action, with respect to such order as they may deem expedient. Nothing in the Master Declaration contained shall be deemed or construed to authorize or permit, by reason of any call of a meeting of Bondowners or of any right conferred hereunder to make such a call, any hindrance or delay in the exercise of any rights conferred upon or reserved to the Paying Agent or the Bondowners under any of the provisions of the Master Declaration.

#### Section 13. Amendment of Master Declaration

**13.1.** Amendment of Master Declaration Without Bondowner Consent. Except during any period during which a Bondowners Committee has been duly constituted and is functioning pursuant to Section 11.5 of this Master Declaration, this Master Declaration may be amended by Supplemental Declaration without the consent of any Bondowners for any one or more of the following purposes:

13.1.1. To cure any ambiguity or formal defect or omission in this Master Declaration;

13.1.2. To add to the covenants and agreements of the District in this Master Declaration, other covenants and agreements to be observed by the District which are not contrary to or inconsistent with this Master Declaration as theretofore in effect;

13.1.3. To authorize issuance of Bonds or Subordinate Obligations;

13.1.4. To authorize Parity Derivative Products, and specify the rights and duties of the parties to a Parity Derivative Product;

13.1.5. To modify, amend or supplement this Master Declaration or any Supplemental Declaration to qualify this Master Declaration under the Trust Indenture Act of 1939, as amended, or any similar federal statute hereafter in effect or to permit the qualification of any Bonds for sale under the securities laws of any of the states of the United States of America;

13.1.6. To confirm, as further assurance, any security interest or pledge created under this Master Declaration or any Supplemental Declaration;

13.1.7. To make any change which, in the reasonable judgment of the District, does not materially and adversely affect the rights of the owners of any Outstanding Bonds;

13.1.8. So long as a Credit Facility (other than a Reserve Credit Facility) is in full force and effect with respect to the Bonds affected by such Supplemental Declaration, to make any other change which is consented to in writing by the issuer of such Credit Facility other than any change which:

13.1.8.1. would result in a downgrading or withdrawal of the rating then assigned to the affected Bonds by the Rating Agencies; or

13.1.8.2. changes the maturity (except as permitted herein), the Interest Payment Dates, interest rates, redemption and purchase provisions, and provisions regarding notices of redemption and purchase applicable to the affected Bonds or diminishes the security afforded by such Credit Facility; or

13.1.8.3. materially and adversely affects the rights and security afforded to the Owners of any Outstanding Bonds not secured by such Credit Facility.

13.1.8.4. To modify any of the provisions of this Master Declaration or any Supplemental Declaration in any other respect whatever, as long as the modification shall take effect only after all affected Outstanding Bonds cease to be Outstanding.

**13.2. Amendment With Bondowner Consent.** Except during any period during which a Bondowners Committee has been duly constituted and is functioning pursuant to Section 11.5 of this Master Declaration, this Master Declaration may be amended for any other purpose only upon consent of Bondowners of not less than fifty-one percent 51% in aggregate principal amount of the Bonds outstanding; provided, however, that no amendment shall be valid without the consent of Bondowners of 100 percent of the aggregate principal amount of the Bonds outstanding which:

13.2.1. Extends the maturity of any Bond, reduces the rate of interest upon any Bond, extends the time of payment of interest on any Bond, reduces the amount of principal payable on any Bond, or reduces any premium payable on any Bond, without the consent of the affected Bondowner; or

13.2.2. Reduces the percent of Bondowners required to approve amendatory Master Declarations.

**13.3. Initial Purchaser As Owner.** For purposes of any consents of Owners required by this Master Declaration or any Supplemental Declaration, including but not limited to Section 13.2, but subject to Section 13.4, the initial purchaser of a series of Bonds may be treated as the Owner of that Series at the time that series of Bonds is delivered in exchange for payment.

**13.4. Issuer of Credit Facility As Owner.** Except as otherwise expressly provided in a Supplemental Declaration, as long as a Credit Facility (other than a Reserve Credit Facility) securing all or a portion of any Outstanding Bonds is in effect, the issuer of such Credit Facility shall be deemed to be the Bondowner of the Bonds secured by such Credit Facility:

13.4.1. at all times for the purpose of the execution and delivery of a Supplemental Declaration or of any amendment, change or modification of this Master Declaration or the initiation by Bondowners of any action which under this Master Declaration requires the written approval or consent of or can be initiated by the Bondowners of at least a majority in principal amount of the affected Bonds at the time Outstanding; and following an Event of Default for all other purposes.

13.4.2. Notwithstanding the foregoing, the issuer of such Credit Facility shall not be deemed to be a Bondowner secured thereby with respect to any such Supplemental Declaration or of any amendment, change or modification of this Master Declaration which:

13.4.2.1. would result in a downgrading or withdrawal of the rating then assigned to the affected Bonds by the Rating Agencies; or

13.4.2.2. changes the maturity (except as expressly permitted herein), the Interest Payment Dates, interest rates, redemption and purchase provisions, and provisions regarding notices of redemption and purchase applicable to the affected Bonds or diminishes the security afforded by such Credit Facility; or

13.4.2.3. reduces the percentage or otherwise affects the classes of affected Bonds, the consent of the Bondowners of which is required to effect any such modification or amendment.

13.4.3. In addition and notwithstanding the foregoing, no issuer of a Credit Facility given as security for any Bonds shall be entitled to exercise any rights under this Section during any period where:

13.4.3.1. the Credit Agreement or Credit Facility to which such Credit Provider is a party shall not be in full force and effect;

13.4.3.2. such Credit Provider shall have filed a petition or otherwise sought relief under any federal or state bankruptcy or similar law;

13.4.3.3. such Credit Provider shall, for any reason, have failed or refused to honor a proper demand for payment under such Credit Facility; or

13.4.3.4. an order or decree shall have been entered, with the consent or acquiescence of such Credit Provider, appointing a receiver or receivers or the assets of the Credit Provider, or if such order or decree having been entered without the consent or acquiescence of such Credit Provider, shall not have been vacated or discharged or stayed within ninety (90) days after the entry thereof.

13.4.4. For purposes of determining the percentage of Bondowners consenting to, waiving or otherwise acting with respect to any matter that may arise under this Master Declaration, the Owners of Bonds which pay interest only at maturity, and mature more than one year after they are issued shall be treated as Owners of Bonds in an aggregate principal amount equal to the accreted value of such Bonds as of the date the Registrar sends out notice of requesting consent, waiver or other action as provided herein.

#### Section 14. Defeasance

The District may defease and deem all or any portion of the Outstanding Bonds to be paid by:

**14.1. Irrevocable Deposit of Direct Obligations.** Irrevocably depositing cash or noncallable, nonprepayable Direct Obligations in escrow with an independent escrow agent which are

calculated to be sufficient, without reinvestment, for the payment of Bonds which are to be defeased; and,

**14.2. Opinion of Qualified Consultant. Filing** with the escrow agent an opinion from a Qualified Consultant to the effect that the money and the principal and interest to be received from the Direct Obligations are calculated to be sufficient, without further reinvestment, to pay the defeased Bonds when due; and,

**14.3. Opinion of Bond Counsel.** Filing with the escrow agent an opinion of nationally recognized bond counsel that the proposed defeasance will not cause interest on the defeased Bonds to be includable in gross income under the Code.

If Bonds are defeased under this Section, all obligations of the District with respect to those defeased Bonds shall cease and terminate, except for the obligation of the District, the escrow agent and the Registrar to pay the defeased Bonds from the amounts deposited in escrow, and the obligation of the Registrar to continue to transfer bonds as provided in this Master Declaration.

#### Section 15. BEO System

**15.1.** Unless otherwise provided by a Supplemental Declaration, all Bonds shall be subject to the BEO System pursuant to the provisions of this Section 15.

15.2. The Bonds shall be initially issued as a BEO security issue with no Bonds being made available to the Owners upon the execution and delivery of the letter of representations among the Paying Agent, DTC and the District. Ownership of the Bonds shall be recorded through entries on the books of banks and broker-dealer participants and correspondents that are related to entries on the DTC BEO system. The Bonds shall be initially issued in the form of separate single fully registered typewritten Bonds for each maturity of the Bonds (the "Global Bonds") in substantially the form attached hereto as Exhibit A with such changes as the Authorized Officer may approve. Each Global Bond shall be registered in the name of CEDE & CO. as nominee (the "Nominee") of DTC (DTC and any other qualified securities depository designated by the District as a successor to DTC, collectively the "Depository") as the "Registered Owner", and such Global Bonds shall be lodged with the Depository until early redemption or maturity of the Bond issue. The Paying Agent shall remit payment for the maturing principal and interest on the Bonds to the Owner for distribution by the Nominee for the benefit of the owners (the "Beneficial Owner" or "Record Owner") by recorded entry on the books of the Depository participants and correspondents. While the Bonds are in BEO form, the Bonds will be available in denominations of \$5,000 or any integral multiple thereof within a maturity.

**15.3.** In the event the Depository determines not to continue to act as securities depository for the Bonds, or the District determines that the Depository shall no longer so act, then the District will discontinue the BEO system with the Depository. If the District fails to designate another qualified securities depository to replace the Depository or elects to discontinue use of a BEO system, the Bonds shall no longer be a BEO issue but shall be registered in the registration books maintained by the Paying Agent in the name of the Owner as appearing on the Bond register and thereafter in the name or names of the Owners of the Bonds transferring or exchanging Bonds.

**15.4.** While the Bonds are in BEO form, the District and the Paying Agent shall have no responsibility or obligation to any participant or correspondent of the Depository or to any Registered Owner on behalf of which such participants or correspondents act as agent for the Owner with respect to:

15.4.1. The accuracy of the records of the Depository, the Nominee or any participant or correspondent with respect to any ownership interest in the Bonds;

15.4.2. The delivery to any participant or correspondent or any other person, other than an Owner as shown in the registration books maintained by the Paying Agent, of any notice with respect to the Bonds, including any notice of prepayment;

15.4.3. The selection by the Depository of the beneficial interest in Bonds to be redeemed prior to maturity; or

15.4.4. The payment to any participant, correspondent, or any other person other than the owner of the Bonds as shown in the registration books maintained by the Paying Agent, of any amount with respect to principal of or interest on the Bonds.

**15.5.** Notwithstanding the BEO system, the District may treat and consider the Owner in whose name each Bond is registered in the registration books maintained by the Paying Agent as the Owner and absolute owner of such Bond for the purpose of payment of principal and interest with respect to such Bond, or for the purpose of giving notices of redemption and other matters with respect to such Bond, or for the purpose of registering transfers with respect to such Bond, or for the purpose of registering transfers with respect to such Bond, or for the purpose of registering transfers with respect to such Bond, or for the purpose of registering transfers with respect to such Bond, or for all other purposes whatsoever. The District shall pay or cause to be paid all principal and interest on the Bonds only to or upon the order of the Registered Owner, as shown in the registration books maintained by the Paying Agent, or their respective attorneys duly authorized in writing, and all such payments shall be valid and effective to fully satisfy and discharge the District's obligation with respect to payment thereof to the extent of the sum or sums so paid.

**15.6.** Upon delivery by the Depository to the District and to the Owner of written notice to the effect that the Depository has determined to substitute a new nominee in place of the Nominee, then the word "Nominee" in this Master Declaration shall refer to such new nominee of the Depository, and upon receipt of such notice, the District shall promptly deliver a copy thereof to the Paying Agent. The Depository shall tender the Bonds it holds to the Paying Agent for reregistration.

#### Section 16. Redemption of Bonds

**16.1.** Unless otherwise provided by a Supplemental Declaration, all Bonds shall be subject to the redemption terms of this Section 16.

16.2. The District reserves the right to purchase Bonds in the open market.

**16.3.** If Bonds are subject to mandatory redemption the Paying Agent shall, without further action by the District, select the particular Bonds to be redeemed in accordance with the mandatory redemption schedule, by lot within each maturity, call the selected Bonds, and give notice of their redemption in accordance with this Section 16.

**16.4.** If certain maturities of Bonds are subject to both optional and mandatory redemption, the District may elect to apply any of those Bonds which it has previously optionally redeemed. In addition, if the District purchases Bonds which are subject to mandatory redemption, the District may elect to apply against the mandatory redemption requirement any such Bonds which it has previously purchased. If the District makes such an election, it shall notify the Paying Agent not less than sixty days prior to the mandatory redemption date to which the election applies.

**16.5.** So long as the BEO-System remains in effect with respect to the Bonds, and unless DTC consents to a shorter period, the Paying Agent shall provide not less than 20 days nor more than 60 days' notice of redemption, and shall provide such information in connection therewith as required by the letter of representations submitted to DTC in connection with the issuance of the Bonds.

**16.6.** During any period in which the BEO System is not in effect with respect to the Bonds, unless waived by any Owner of the Bonds to be redeemed, official notice of any redemption of Bonds shall be given by the Paying Agent on behalf of the District by mailing a copy of an official redemption notice by first class mail postage prepaid at least 3days and not more than 60 days prior to the date fixed for redemption to the Owner of the Bonds to be redeemed, at the address shown on the Revenue Bond Register or at such other address as is furnished in writing by such owner to the Paying Agent. All such official notices of redemption shall be dated and shall state:

16.6.1. The redemption date;

16.6.2. The redemption price;

16.6.3. If less than all Outstanding Bonds are to be redeemed, the identification (and, in the case of partial redemption, the respective principal amounts) of the Bonds to be redeemed;

16.6.4. That on the redemption date the redemption price will become due and payable upon each such Bond or portion thereof called for redemption, and that interest thereon shall cease to accrue from and after said date; and

16.6.5. The place where such Bonds are to be surrendered for payment of the redemption price, which place of payment shall be the principal office of the Paying Agent.

**16.7.** The District shall deposit with the Paying Agent, on or before the redemption date, an amount of money sufficient to pay the redemption price of all the Bonds or portions of Bonds which are to be redeemed on that date.

**16.8.** Official notice of redemption having been given as aforesaid, the Bonds or portions of Bonds to be redeemed shall, on the redemption date, become due and payable at the Redemption Price therein specified, and from and after such date (unless the District shall default in the payment of the redemption price or unless the notice was conditional as described in Section 15.9) such Bonds or portions of Bonds shall cease to bear interest. Upon surrender of such Bonds for redemption in accordance with said notice, such Bonds shall be paid by the Paying Agent at the Redemption Price. Installments of interest due on or prior to the redemption date shall be payable as herein provided for payment of interest. Upon surrender for any partial

redemption of any Bond, there shall be prepared for the registered owner a new Bond of the same maturity in the amount of the unpaid principal. All Bonds which have been redeemed shall be canceled and destroyed by the Paying Agent and shall not be reissued. Notwithstanding that any Bonds called for redemption shall not have been surrendered, no further interest shall accrue on any such Bonds. From and after such notice having been given and such deposit having been made, the Bonds to be redeemed shall not be deemed to be Outstanding hereunder, and the District shall be under no further liability in respect thereof.

**16.9.** Any notice of optional redemption given for the Bonds pursuant to this Section 16 may state that the optional redemption is conditional upon receipt by the Paying Agent of moneys sufficient to pay the redemption price of such Bonds or upon the satisfaction of any other condition, and/or that such notice may be rescinded upon the occurrence of any other event, and any conditional notice so given may be rescinded at any time before payment of such redemption price if any such condition so specified is not satisfied or if any such other event occurs. Notice of such rescission or of the failure of any such condition shall be given by the Paying Agent to affected owners of the Bonds as promptly as practicable.

## Section 17. Authentication, Registration and Transfer

**17.1.** The provisions of this Section 17 apply only if the Bonds cease to be a BEO issue, and unless otherwise specified in a Supplemental Declaration.

**17.2.** No Bond shall be entitled to any right or benefit under this Master Declaration unless it shall have been authenticated by an authorized officer of the Paying Agent. The Paying Agent shall authenticate all Bonds to be delivered at Closing, and shall additionally authenticate all Bonds properly surrendered for exchange or transfer pursuant to this Master Declaration.

**17.3.** All Bonds shall be in registered form. [\_\_\_\_\_] is hereby appointed to serve as Paying Agent for the Bonds. A successor Paying Agent may be appointed for the Bonds by ordinance or resolution of the District. The Paying Agent shall provide notice to Owners of any change in the Paying Agent not later than the Bond payment date following the change in Paying Agent.

**17.4.** The ownership of all Bonds shall be entered in the Bond register maintained by the Paying Agent and the District and Paying Agent may treat the person listed as owner in the Bond register as the owner of the Bond for all purposes.

**17.5.** The Paying Agent shall mail each interest payment on the Interest Payment Date (or the next Business Day if the Interest Payment Date is not a Business Day) to the name and address of the Owner, as that name and address appear on the Bond register as of the Record Date. If payment is so mailed, neither the District nor the Paying Agent shall have any further liability to any party for such payment.

**17.6.** Bonds may be exchanged for an equal principal amount of Bonds of the same Series and maturity which are in different authorized denominations, and Bonds may be transferred to other owners if the Owner submits the following to the Paying Agent:

17.6.1. Written instructions for exchange or transfer satisfactory to the Paying Agent, signed by the Owner or his attorney in fact and guaranteed or witnessed in a manner satisfactory to the Paying Agent; and

17.6.2. The Bonds to be exchanged or transferred.

**17.7.** The Paying Agent shall not be required to exchange or transfer any Bonds submitted to it during any period beginning with a Record Date and ending on the next following payment date; however, such Bonds shall be exchanged or transferred promptly following the payment date.

**17.8.** The Paying Agent shall not be required to exchange or transfer any Bonds which have been designated for redemption if such Bonds are submitted to it during the fifteen-day period preceding the designated redemption date.

**17.9.** For purposes of this Section, Bonds shall be considered submitted to the Paying Agent on the date the Paying Agent actually receives the materials described in Section 17.6.

**17.10.** The District may alter these provisions regarding registration and transfer by mailing notification of the altered provisions to all Owners. The altered provisions shall take effect on the date stated in the notice, which shall not be earlier than 45 days after notice is mailed.

## Section 18. The Series 2016 Bonds

**18.1.** Pursuant to the authority of Order \_\_\_\_\_ and this Master Declaration, the District has issued its [Sewer System Revenue Bonds, Series 2016], in the aggregate principal amount of \$\_\_\_\_\_. The Series 2016 Bonds shall be Bonds as defined in this Master Declaration. The Series 2016 Bonds shall bear interest payable on \_\_\_\_\_\_ and \_\_\_\_\_ of each year at the following rates, commencing \_\_\_\_\_\_, 2016, and shall mature in the following years in the following principal amounts:

Due	Principal <u>Amount</u>	Interest <u>Rate</u>	CUSIP

\* Subject to mandatory sinking fund redemption as described in the next subsection.

**18.2.** The Series 2016 Bonds maturing on \_\_\_\_\_\_, are subject to mandatory sinking fund redemption at a price of 100 percent of the principal amount of the Series 2016 Bonds to be redeemed plus accrued interest to the date of redemption on June 1 in years and amounts as follows:

Date Mandatory \_\_\_\_\_ Sinking Fund Redemption

\* Final maturity.

A Series 2016 Term Bond subject to optional redemption and redeemed in part will have the principal amount within the respective mandatory redemption dates selected by the District.

**18.3.** The Series 2016 Bonds maturing on or after \_\_\_\_\_ may be redeemed at the option of the District on or after \_\_\_\_\_, at par plus accrued interest to the date of redemption.

**18.4.** The Series 2016 Bonds shall be special obligations of the District, and shall be payable solely from the Net Revenues and amounts required to be deposited in the Revenue Bond Account and Revenue Bond Reserve Account as required and as provided by this Master Declaration. The Series 2016 Bonds are not general obligations of the District and are payable solely from the amounts described in the previous sentence.

**18.5.** The Series 2016 Bonds shall be in substantially the form attached as Appendix A and shall be signed with the facsimile or manual signature of an Authorized Officer.

18.5.1. [The Series 2016 Bond proceeds received by the District shall be applied as follows:

18.5.1.1. \$\_\_\_\_\_\_ to the First Reserve Subaccount or to acquire a Reserve Credit Facility for the First Reserve Subaccount to meet the First Reserve Subaccount Reserve Requirement;

18.5.1.2. \$\_\_\_\_\_\_\_ to refund borrowings issued under the District's Master Sewer Revenue Bond Order dated September 15, 1994 and [describe DEQ loans that are refunded]; and

18.5.1.3. The balance of the proceeds to pay costs incurred in connection with the issuance of the Series 2016 Bonds.]

#### Section 19. Tax-Exempt Status

**19.1. General Covenant.** The District covenants for the benefit of the Owners of the Series 2016 Bonds to comply with all provisions of the Code which are required for interest on the Series 2016 Bonds to be excluded from gross income for federal taxation purposes. In

determining what actions are required to comply, the District may rely on an opinion of Bond Counsel. The District makes the following specific covenants with respect to the Code:

19.1.1. The District will not take any action or omit any action if it would cause the Series 2016 Bonds to become "arbitrage Bonds" under Section 148 of the Code.

19.1.2. The District shall operate the facilities financed with the Series 2016 Bonds so that the Series 2016 Bonds do not become private activity Bonds within the meaning of Section 141 of the Code.

19.1.3. The District shall pay, when due, all rebates and penalties with respect to the Series 2016 Bonds which are required by Section 148(f) of the Code.

**19.2.** Covenants as Contracts. The covenants contained in Section 19 and any covenants in the closing documents for the Series 2016 Bonds shall constitute contracts with the owners of the Series 2016 Bonds, and shall be enforceable by them.

EXECUTED ON BEHALF OF THE DISTRICT BY ITS AUTHORIZED OFFICER AS OF THE \_\_\_\_\_ DAY OF \_\_\_\_\_, 2016.

#### CLACKAMAS COUNTY SERVICE DISTRICT NO. 1

By: \_\_\_\_

AUTHORIZED OFFICER

#### Appendix A

No. R-«BondNumber»

United States of America State of Oregon County of Clackamas Clackamas County Service District No. 1 [Sewer Revenue and Refunding Bond] Series 2016

 Dated Date:
 [closing date], 2016

 Interest Rate Per Annum:
 «CouponRate»%

 Maturity Date:
 \_\_\_\_\_\_1, «MaturityYear»

 CUSIP Number:
 179282«CUSIPNumbr»

 Registered Owner:
 -----Cede & Co.---- 

 Principal Amount:
 -----«PrincipalAmtSpelled» Dollars---- 

CLACKAMAS COUNTY SERVICE DISTRICT NO. 1, in Clackamas County, Oregon (the "District"), for value received, acknowledges itself indebted and hereby promises to pay to the Registered Owner hereof, or registered assigns, but solely from the sources indicated below, the Principal Amount indicated above on the Maturity Date indicated above together with interest thereon from the date hereof at the Interest Rate Per Annum indicated above. Interest is payable semiannually on the \_\_\_\_\_ day of \_\_\_\_\_\_ and the \_\_\_\_\_ day of

\_\_\_\_\_\_ in each year until maturity or prior redemption, commencing \_\_\_\_\_\_ 1, 20\_\_. Payment of each installment of principal or interest shall be made to the Registered Owner hereof whose name appears on the registration books of the District maintained by the District's registrar and paying agent, which is currently [Paying Agent and Registrar] (the "Registrar") as of the close of business on the fifteenth day of the calendar month immediately preceding the applicable interest payment date. For so long as this Bond is subject to a book-entry-only system, principal and interest payments shall be payable in same-day funds on each payment date to the nominee of the securities depository for the Bonds. On the date of issuance of this Bond, the securities depository for the Bonds is The Depository Trust Company, New York, New York ("DTC"), and Cede & Co. is the nominee of DTC. Such payments shall be made payable to the order of "Cede & Co."

This Bond is one of a duly authorized series of bonds of the District aggregating \$[principal amount] in principal amount and designated as [Sewer Revenue and Refunding Bonds], Series 2016 (the "Bonds"). The Bonds are issued as provided in the District's Master Sewer Revenue Bond Declaration dated [closing date], and the District's Order No. \_\_\_\_\_\_\_, 2016 (collectively, the "Master Declaration"), and under the authority of and in full compliance with the Constitution and Statutes of the State of Oregon. Capitalized terms used but not defined herein shall have the meanings assigned them in the Master Declaration.

This Bond is not a general obligation of the District, and is payable solely from the Net Revenues of Sewer System of the District, as provided in the Master Declaration. The Net Revenues have been pledged to pay the Bonds.

The Bonds are initially issued as a book-entry-only security issue with no certificates provided to the Owners. Records of Bond ownership will be maintained by DTC and its participants.

Should the book-entry-only security system be discontinued, the Bonds shall be issued in the form of registered Bonds without coupons in denominations of \$5,000 or any integral multiple thereof. Such Bonds may be exchanged for bonds of the same aggregate principal amount, but different authorized denominations, as provided in the Master Declaration.

The Bonds mature and are subject to redemption as described in the final Official Statement for the Bonds.

Any exchange or transfer of this Bond must be registered, as provided in the Master Declaration, upon the bond register kept for that purpose by the Registrar. The exchange or transfer of this Bond may be registered only

Page 40 – Appendix A – Form of Bond

\$«PrincipalAmtNumber»

by surrendering it, together with a written instrument of exchange or transfer which is satisfactory to the Registrar and which is executed by the registered owner or duly authorized attorney. Upon registration, a new registered Bond, of the same series and maturity and in the same aggregate principal amount, shall be issued to the transferee as provided in the Master Declaration. The District and the Registrar may treat the person in whose name this Bond is registered on the bond register as its absolute owner for all purposes, as provided in the Master Declaration.

Unless this certificate is presented by an authorized representative of DTC to the District or the Registrar for registration of transfer, exchange or payment, and any certificate issued is registered in the name of Cede & Co. or such other name as is requested by an authorized representative of DTC (and any payment is made to Cede & Co. or to such other entry as is requested by an authorized representative of DTC), ANY TRANSFER, PLEDGE OR OTHER USE HEREOF FOR VALUE OR OTHERWISE BY OR TO ANY PERSON IS WRONGFUL inasmuch as the registered owner hereof, Cede & Co., has an interest herein.

This Bond shall remain in the Registrar's custody subject to the provisions of the FAST Balance Certificate Agreement currently in effect between the Registrar and DTC.

It is hereby certified, recited and declared that all acts, conditions and things required by the Constitution and Statutes of the State of Oregon to exist, to have happened and to have been performed precedent to and in the issuance of this Bond to exist, have happened and have been performed in due time, form and manner as prescribed by law, and that the amount of this Bond, together with all other obligations or indebtedness of the District, does not exceed any constitutional or statutory limitations of indebtedness.

IN WITNESS WHEREOF, Clackamas County Service District No. 1, Oregon, has caused this Bond to be executed with the facsimile signatures of the District Official, all as of the date indicated above.

#### **Clackamas County Service District No. 1, Oregon**

Authorized Officer

## THIS BOND SHALL NOT BE VALID UNLESS PROPERLY AUTHENTICATED BY THE REGISTRAR IN THE SPACE INDICATED BELOW.

#### **CERTIFICATE OF AUTHENTICATION**

This Bond is one of the \$[principal amount] aggregate principal amount of Clackamas County Service District No. 1 [Sewer Revenue and Refunding Bonds], Series 2016, and is authorized to be issued under the terms of the Master Declaration described herein.

Date of authentication: [closing date], 2016.

[Paying Agent and Registrar], as Registrar

Authorized Officer

\*\*\*\*\*

Page 41 – Appendix A – Form of Bond

according a chorage many

#### ASSIGNMENT

FOR VALUE RECEIVED, the undersigned sells, assigns and transfers unto: \_

(Please insert social security or other identifying number of assignee)

this Bond and does hereby irrevocably constitute and appoint \_

as attorney to transfer this Bond on the books kept for registration thereof with the full power of substitution in the premises.

Dated:

NOTICE: The signature to this assignment must correspond with the name of the registered owner as it appears upon the face of this Bond in every particular, without alteration or enlargement or any change whatever.

NOTICE: Signature(s) must be guaranteed by a member of the New York Stock Exchange or a commercial bank or trust company

Signature Guaranteed

(Bank, Trust Company or Brokerage Firm)

Authorized Officer

The following abbreviations, when used in the inscription on the face of this Bond, shall be construed as though they were written out in full according to applicable laws or regulations.

TEN COM -- tenants in common TEN ENT -- as tenants by the entireties JT TEN -- as joint tenants with right of survivorship and not as tenants in common OREGON CUSTODIANS use the following: CUST UL OREG MIN

as custodian for (name of minor) OR UNIF TRANS MIN ACT under the Oregon Uniform Transfer to Minors Act

Additional abbreviations may also be used though not in the list above.

\*\*\*\*\*\*\*

\*\*\*\*\*

Page 42 – Appendix A – Form of Bond

COACIA CALORAE DODIE